

Received 12 January 2024, accepted 7 February 2024, date of publication 9 February 2024, date of current version 16 February 2024. Digital Object Identifier 10.1109/ACCESS.2024.3364676

TOPICAL REVIEW

Drone Technologies: A Tertiary Systematic Literature Review on a Decade of Improvements

JURRIAN DOORNBOS[®]¹, KWABENA E. BENNIN¹, ÖNDER BABUR[®]^{1,2}, AND JOÃO VALENTE^(D), (Member, IEEE)

¹Information Technology Group, Wageningen University & Research, 6700 HB Wageningen, The Netherlands
²Department of Mathematics and Computer Science, Eindhoven University of Technology, 5612 AZ Eindhoven, The Netherlands

Corresponding author: Jurrian Doornbos (jurrian.doornbos@wur.nl)

This work was supported by the European Commission in the Scope of its Horizon Europe Program, Innovation and Capacity building in Agricultural Environmental and Rural Uav Services (ICAERUS: https://icaerus.eu/) under Contract 101060643.

ABSTRACT Unmanned aerial vehicles (UAVs) have emerged as versatile tools with significant potential in various fields, including, but not limited to civil engineering, ecology, networking and precision agriculture. Systematic literature reviews (SLRs) play a crucial role in assessing the quality of research methods and approaches, aiding researchers and practitioners in selecting and optimizing their projects. However, the quality assessment of UAV-related SLRs and the aggregation of UAV technologies across research fields remain limited. This study aims to address these gaps by conducting a tertiary literature review (TLR) that assesses the quality of SLRs, aggregates data across research fields, and provides guidelines for researchers and practitioners in the UAV community. Based on a review of 73 SLRs, it is evident that the quality of UAV-related SLRs is generally low, with a lack of quality assessment and inadequate reporting of detailed information on primary studies. Consequently, this study presents reporting items and example quality assessment ratings sourced from the SLRs to enhance the transparency and comparability of future UAV-related research. Additionally, it highlights common limitations faced by UAV applications, such as regulatory, technical, social, and research-related challenges, which require attention for progress in the field. Overall, this study aims to enhance the quality and knowledge sharing in the UAV research community by providing insights into the current state of UAV-related SLRs and offering practical guidance for researchers and practitioners through the provision of data-extraction templates and quality control questions for future UAV-related reviews and primary research.

INDEX TERMS Quality assessment, systematic literature review (SLR), tertiary literature review (TLR), UAV payloads, UAV platforms, UAV software, unmanned aerial vehicles (UAVs).

I. INTRODUCTION

The past decade has seen the arrival and proliferation of commercial drones. In these first years, drones were experimentally tested for various tasks such as agricultural crop detection, search-and-rescue operations, and medical goods transportation. However, decreasing costs and simpler operations made Unmanned Aerial Vehicles (UAVs) an increasingly common tool for data acquisition, transport, and

The associate editor coordinating the review of this manuscript and approving it for publication was Jiankang Zhang¹⁰.

real-time operations across industry, the public sector, and more academic niches.

This widespread adoption of UAVs has created a great diversity in UAV users. Each of these practitioners has created their own solution to apply UAV technology to their particular use case. For example, in a review on UAVs for supporting algal bloom research, [1] identified 10 unique sensors on 10 different flying platforms, used in four different ecosystems, with the use of seven different spectral indices. Whereas in a review on UAVs for parcel and passenger transportation [2], six different problems facing UAVs for

transportation and three different solutions were presented. Furthermore, in another review on UAVs for forest firerelated usage [3], eight different UAV characteristics for fire extinguishing tasks, ranging from UAV group size to fire-extinguishing tools and collision avoidance algorithms, covering 51 articles were found. These examples demonstrate that for each use case, depending on the UAV model and payload, unique solutions are created [1].

These solutions require a high number of resources to develop, raising the barrier to entry [4]. For example, UAVs can be more discrete than humans for monitoring vulnerable species [5], although complex sensors and data processing pipelines need to be developed for the best results [6]. In addition, developing countries with high forest cover have very little published UAV research [7], even though these forests have the highest need for careful monitoring programs. Another example is in healthcare, where UAV transport of medical goods is often faster than the standard response, however, a medical payload also requires a clinical and sterile container, secured onto the UAV [8], which increases research and development costs. Lowering the barrier to entry can therefore be supported by identifying which technologies, practices and methods have been more prominent in research.

An important method in sharing these technologies, practices, and methods is a Systematic Literature Review [9] (SLR). SLRs originated from medical sciences and were adapted for Software Engineering by [10]. This adaptation presents a systematic methodology for searching articles, extracting information, and rating the study using quality assessment criteria.

A review of papers using the SLR method was published by the same group three years later [9]. Their findings indicate that the literature reviews are not in accordance with any method, many do not perform a quality assessment, and only a few of their included articles present guidelines for practitioners, which is one of the primary benefits of a literature review. This 'review of reviews' is often called a tertiary literature review (TLR). Different from an SLR, a TLR aggregates information only from secondary studies [9]. A TLR is performed to consolidate information across fields; identify research topics; aggregate scattered data; find limitations and assess the quality of SLRs with the DARE rating scheme [11]. In [12] for example, the TLR method from [9] was adapted to Machine Learning in Software Engineering. The impact of Machine Learning on the field was assessed and a classification scheme for categorizing applications was presented. Additionally, the quality assessment of review articles indicates an average rating of 3 out of 4, using the DARE rating scheme [11] (more on this scheme will be presented in Section III-C).

To the best of our knowledge, only a single TLR has been published within the UAV field. The TLR from [13] covers UAV use in civil engineering. They identified concerns for UAV deployment and future directions for using UAVs in civil engineering. More problematic however is that the results of their quality assessment on UAV reviews show low results: on average the reviews got a quality rating of 0.3 out of 3 [13]. Only 3 quality criteria were measured from the DARE rating scheme in the TLR. The authors indicate that the low score is mainly from a shortage of quality assessment being performed in their included SLRs. Additionally, the TLR did not aggregate data on UAVs across research fields.

SLRs are a valuable tool to assess the quality of methods and approaches, helping researchers and practitioners select, and optimize their projects [10]. However, when the quality assessment is not performed, or an aggregation of data is not executed, researchers and practitioners miss out on key information to reduce the barrier to entry for using UAV technology [4]. To support research and practitioners of UAV technology, an understanding and improvement of SLR quality is needed in addition to a broad aggregation of UAV technologies used across research fields. The aim of this study is therefore threefold. We aim to:

- Understand the quality of UAV-related SLRs,
- Provide a broad aggregation of UAV technologies, and
- Present further steps in lowering the barrier to entry for research and practitioners of UAV technology.

In this study. a tertiary literature review is conducted. This TLR assesses the quality of SLRs, aggregates data across research fields, and can provide broad guidelines to be applicable across the widespread UAV research community. This aim is focused on providing items for practitioners and researchers, which is reflected in the main research question:

How can future UAV studies be conducted to improve the quality for optimizing information sharing for practitioners and researchers?

This question consists of the following sub-questions:

- RQ1. What is the quality of systematic literature reviews conducted in the UAV domain?
- RQ2. What are the popular UAV platforms, payloads, and software in the UAV domain?
- RQ3. Which limitations for using UAVs are identified in the SLRs?

Our TLR on 73 SLRs reveals that the quality of SLRs on UAVs is not high: SLRs are not performing a quality assessment step. Additionally, SLRs do not report detailed information on their primary studies, making comparative analysis difficult. Therefore, various reporting items and example quality assessment ratings, sourced from the SLRs, are presented in the supplement of this article for primary and secondary research. Finally, many UAV applications are facing similar limitations, solving these regulatory, technical, social, and research-related limitations will be a fruitful endeavor in the coming years.

The main contributions of this review are as follows: 1. providing a database of UAV reviews and UAV technologies, 2. assessing the quality of reviews, and 3. publishing data-extraction templates and quality control questions for future UAV-related reviews and UAV primary research. The article is structured as follows: Section II covers related work, Section III presents the TLR methodology, Section IV presents the results, Section V presents the discussion, Section VI identifies threats to the validity of this study and Section VII presents the conclusion.

II. BACKGROUND AND RELATED WORK

The technology for UAVs has a background in military usage. The design of large aircraft such as the NASA Ikhana and Sierra were created with high-altitude surveillance or even tactical operations in mind [14]. Over time, microchips became faster and smaller, and battery life and power output increased, making smaller and more affordable UAVs available to a wider audience. These smaller aircraft were the main subject of early work in using UAVs for various non-military use cases which at the time were classified as small, mini, micro, and nano-sized UAVs [14]. The overview of different UAV platforms and the operational range provided is presented as follows: large UAVs have an operating range of approximately 500 km and can fly for up to 2 days; medium UAVs have a similar operating range but a shorter flight time of around 10 hours; small UAVs have an operating range of fewer than 10 km and a flight time of fewer than 2 hours; mini UAVs can fly with up to 5 kg and typically require hand-launching; micro and nano UAVs have even more limited operating ranges, and have very short flight times of less than 1 hour.

In the article from [14], the authors argue that these lightweight (small, mini, micro, and nano) UAVs will revolutionize spatial ecology. The birds-eye perspective, in combination with the high-resolution sensors and user-controlled revisit periods, is posed to fill the gap of data required by spatial ecology, at an affordable cost to many researchers. They present three usages of UAVs for spatial ecology monitoring: 1. population ecology through thermal sensing; 2. vegetation dynamics through multi and hyperspectral sensors; and 3. ecosystem processes through miniature temperature, pressure, and particle sensors.

A similarly influential work by [15] reviews the use of UAVs for photogrammetry and remote sensing. Their classification extends previous works by separating fixed-wing and multi-rotor aircraft within the lightweight categories. They provide an overview of various technologies that make up a UAV system. The overview presented by the authors starts with the flying platform itself, with the various classifications and usages, a ground-control station, communication between the various parts of the UAV system, and mission planning. Regulatory bodies and regulations are covered for various regions of the world. These mainly cover working groups, preliminary regulations, and older regulations affecting these new small systems, whilst arguing for the renewal of many of these regulations. Various payloads for the UAV are presented, in the form of autopilots and navigation systems, orientation systems, and sensing payloads. Critically, many examples are given of the available technologies. In the processing section, an overview of image orientation and camera calibration methods is presented and the state-of-the-art for surface reconstruction is shown. They identify that the high spatial resolution that these new UAV platforms create poses a new problem to existing photogrammetry methods: the requirement of having extremely accurate positioning information for accurate image alignment and surface reconstruction methods. They conclude that there is a high variance between UAV studies, therefore the extrapolation of results to other UAV studies has to be made with caution.

In the years since these initial forays, many studies, surveys, and reviews have been performed on the use of UAVs for various applications across academia.

The aforementioned study from [13] is the only tertiary study we identified, which covers 33 secondary reviews on UAVs for civil engineering and construction. Their tertiary study identified three main concerns with UAV deployment: restrictive regulations, system reliability, and factors affecting UAV use. Future directions for UAV studies were directed to the development of sensors, algorithms, navigation, and capabilities. Additionally, they propose more studies into the feasibility, human-UAV interaction, and effectiveness of specific UAV solutions. Finally, the study found that the quality of UAV research reviews is quite low: on average the reviews got a quality score of 0.3 out of 3 [13].

A similar topic of UAV technology and best practices is also covered in the study by [4]. Their focus is on presenting onboard UAV sensors and their best practices for calibration, improving mapping quality, emerging acquisition scenarios, trends in UAV data processing, and future challenges for UAV technology. Each of these topics is covered by providing examples from research or industry. Additionally, in the case of larger research topics, the reader is directed toward review articles. The presented best practices are clear and actionable, usable for many starting and experienced practitioners. Examples include how ground control points should be placed for the best results for photogrammetry, tips on battery life optimization, and how to calibrate a multispectral sensor. Finally, the authors argue that more cross-pollination in UAV research is required, by starting from the perspective that UAV applications are scattered, and that the technology depends heavily on progress in other research fields. Such as increased battery efficiency, computer vision research, and miniaturization of processing power [4].

To complement these prior studies, we conduct a systematic tertiary literature review that will analyze the quality of prior review articles, systematically aggregate data from these reviews and identify a set of common limitations across UAV research. Our study is different from prior studies in that we systematically reflect on the SLR method itself, to provide guidelines for future SLRs and the field as a whole.

III. METHODOLOGY

The Systematic Literature Review method from [10] was adapted for tertiary studies in [9]. These guidelines present a literature review according to three phases: planning,



FIGURE 1. PRISMA inclusion/exclusion criteria chart. Exclusion reasons are presented in columns on the right side: duplicate records, language, not available, and no clear search query in review. Resulting in the 73 included SLRs.

conducting, and reporting. During the planning phase, a search protocol is developed and reviewed by all authors. This protocol consists of a search process, quality assessment using Database of Abstracts for Reviews of Effects (DARE) criteria, and data extraction. The following section presents the protocol followed and the adaptation to UAV applications.

A. THE SEARCH PROCESS

Three databases and one indexing system were searched for articles. The searched databases were: IEEE Xplore, ACM, and Clarivate Web Of Science. The indexing system was Elsevier Scopus. These were searched between August 9th and September 9th. All the databases had the following query for all fields (title, keywords, abstract): "UAV" AND ("review of studies" OR "structured review" OR "systematic review" OR "literature review" OR "literature analysis" OR "literature survey"). The terms could have been extended by using RPAS (remotely piloted aerial system), UAS (unmanned aerial system), and drone. The variations for 'review' were directly adapted from [9].

Furthermore, to acquire all relevant information based on the recent usage of UAVs, the results were limited to 2012 onward. These queries resulted in a total of 608 studies. No backward or forward snowballing was executed, as the three databases and indexing system already showed a large overlap in results, and the search was deemed exhaustive.

B. STUDY SELECTION

The 608 studies were included following a process adapted from the Preferred Reporting Items for Systematic Reviews (PRISMA) statement [16]. The statement explicitly calls for an identification, screening, and inclusion stage, with specific definitions for each step presented in figure 1.

The inclusion criteria were the following:

• Reviews on UAV usage and UAV technologies across all domains were included.

The exclusion criteria were the following:

- · Non-secondary studies were excluded.
- Studies where UAV was not an abbreviation for Unmanned Aerial Vehicles were excluded.
- Articles not written in English were excluded.
- Based on a 'defined search process' were excluded (not reporting search queries and database).
- Inaccessible papers and reports were excluded.

The most important exclusion step for selecting Systematic Literature Reviews was whether the review is 'based on a defined search process', directly taken from [9]. This means that the exact search queries and the searched databases were clearly presented in the article in the form of text or figures. These inclusions led to the final inclusion of 73 systematic literature review articles. These SLRs cover a total of 8536 primary sources, however, overlap will exist between the included primary sources.

C. QUALITY ASSESSMENT

Each review was manually evaluated according to four criteria presented in [9], presented in table 1. These criteria are adopted from DARE [11]. This is a strict review assessment scheme, initially developed for medical literature reviews, but has seen widespread use across academic disciplines [17]. This quality check was executed by the first author. These questions (QA1, QA2, QA3, QA4) were scored (Yes (Y) = 1, Partly (P) = 0.5, No (N) = 0) according to the scheme in table 1.

D. DATA EXTRACTION PROCESS

The data extraction process is presented in figure 2. Quality assessment in figure 2 are the quality assessment ratings as presented in section III-C above. Metadata in figure 2 consisted of the following items:

- Title and source (Journal article, conference proceedings)
- Journal title
- Number of included primary articles
- Review topic (Summary of topic based on the article title)
- Quality assessment score
- · Whether it included grey literature sources
- Search year range (start year and end year)
- Research category (based on the subject matter)
- Whether it contained detailed data extraction on hardware or software

Aggregated findings on popular UAV technologies were subdivided into hardware and software. Naturally, only the SLRs that performed a detailed data extraction provided information for this extraction (n=21). Additionally,

Quality Assessment (QA)	Rating scheme
questions	.
QA1. Are the review's in-	Y, the inclusion criteria are explic-
clusion and exclusion cri-	itly defined in the paper;
teria described and rele-	
vant?	
	P, the inclusion criteria are implicit;
	N, the inclusion criteria are not de-
	fined and cannot be readily inferred.
QA2. Is the literature	Y, the authors have either searched
search likely to have	four or more digital libraries and in-
covered all relevant	cluded additional search strategies
studies?	or identified and referenced all jour-
	nais addressing the topic of interest;
	P, the authors have searched 2 or 3
	digital libraries with no extra search
	strategies, or searched a defined but
	and proceedings:
	N the authors have searched up to
	1 digital library or an extremely re
	stricted set of journals:
0A3 Did the reviewers as-	V the authors have explicitly de-
sess the quality/validity of	fined quality criteria and extracted
the included studies?	them from each primary study:
the mended studies.	P the research question involves
	quality issues that are addressed by
	the study:
	N. no explicit quality assessment
	of individual papers has been at-
	tempted or quality data has been ex-
	tracted but not used;
QA4. Were the basic	Y, Information is presented about
data/studies adequately	each paper so that the data sum-
described?	maries can be traced to relevant pa-
	pers;
	P, only summary information is pre-
	sented about individual papers e.g.
	papers are grouped into categories
	but it is not possible to link individ-
	ual studies to each category;
	N, the results of the individual stud-
	ies are not specified i.e. the individ-
	ual primary studies are not cited;

TABLE 1. Quality Assessment rating scheme. Adapted from [9] and [11].

hardware was split into platform and payload tables. Where the platform is the flying platform, and the payload is the sensor or other mounted component on the platform. This is the result of starting the extraction based on the detailed extraction from [18] and adapting it when other SLRs reported something new, which is a recommended procedure [9], [10]. Limitations were identified by screening the SLRs with a QA score over three (n=16) and searching for limitations of UAV use according to the SLR discussion sections. The reader is encouraged to explore the complete data extraction, including comparative tables, found in Appendix A. Supplement I. data extraction.

Below, we explain additional details relating to how the data extraction was done for each research question.

RQ1. What is the quality of systematic literature reviews conducted in the UAV domain? To elaborate on the SLRs that performed a Quality Assessment, the details on the specific method of Quality Assessment were extracted from these SLRs (n=3).



FIGURE 2. Data extraction flowchart, spreadsheets were used to track extracted variables from the SLRs: (metadata, QA scores, and aggregations: hardware (payload and platform), software, limitations).

RQ2. What are the popular UAV platforms, sensors and software in the UAV domain? The data extraction on UAV technologies as presented in figure 2 had some minor adaptations to be made. For example, some SLRs extracted 'sensor type' and others 'payload model', however, the content is similar. These were all adapted into the same field, called 'payload type'. More importantly, however, a data-cleaning step was included to make variables group-able and comparable. For example payload naming conventions: different spellings for the MicaSense MCA lite Multispectral sensor would all be renamed to: 'MicaSense MCA-Lite'. This became more problematic when an SLR overlooked critical details. For example, a 'DJI Phantom 3' itself does not exist, but a 'DJI Phantom 3 Professional' does exist. Such cases were left as is. An additional grey literature search was performed to identify the key details of each sensor, software and payload. The resulting data extraction tables for this research question are presented as supplementary materials I and II (supplement I. data extraction: UAVDataExtraction[.xlsx/.csv], supplement II. variables explained: UAVDataExtractionSupportingDocument.pdf).

RQ3. Which limitations for using UAVs are identified in the SLRs? The limitations to using UAVs were identified by searching for limitations presented in the discussion section of the SLRs with a quality score over three (n=16). This additional exclusion rule is to limit the total number of articles on which to perform this resource-intensive task and prioritize the findings from higher-quality SLRs, which is promoted as a method by [19]. The limitations were categorized under regulatory, technical, social, and research limitations, as these four categories covered all of the recurring limitations. Based on the extracted limitations, as presented in figure 2, it became clear that future research could be supported with predefined extraction tables. These extraction tables were developed using the aggregated extraction on hardware and software presented for RQ2. The



FIGURE 3. Year of publishing for included articles.

aggregation was initially based on the extraction in [18] and continually adapted to fit the extractions performed in the other SLRs. This results in extraction tables that cover most of the variables that UAV SLRs have reported upon UAV model type, analytical models used, the accuracy of the analytical model, sensor settings, etc.

IV. RESULTS

A. OVERVIEW OF DATA EXTRACTION

In this section, the results from the data extraction are presented. This covers metadata of the included SLRs and their topic categorizations.

The number of published review articles steadily increases over time, as seen in figure 3. Although the search query ranged from 2012 to August of 2022, systematic literature reviews only started to be published from 2016 onward [18], [20]. Even though the search stopped in August of 2022, the year 2022 already contains the highest number of published articles. This increase is in accordance with the review articles, which indicate a steep incline of newly published work [21], [22]. This increase was also expected in the study of [15] as it shows the increasing popularity of the field.

For topics (see section III-D) with more than 1 included review, the growth of individual fields is shown in figure 4. Forestry and Precision Agriculture have seen the biggest increase in systematic literature reviews since 2020. Furthermore, the construction field is also a significant newcomer to publishing SLRs on UAV usage. The rapid increase seems to indicate large growth in future systematic literature reviews in the coming years.

Figure 11 shows that UAV research is active in a wide variety of research fields, ranging from noise pollution from UAVs to forestry, agriculture, traffic monitoring, and networking. Furthermore, figure 11 shows that search ranges are often left as default (not reporting start and end-date) and when they are reported, researchers often set the start date around 2010. This results in some SLRs including UAV research which might be irrelevant to modern usages of UAVs. Additionally, the number of included articles varies highly; some chose to include 3 articles, and others included



FIGURE 4. Topics of included SLRs over time. For visual clarity, the figure only covers the topics with more than one included SLR on the topic.

over 300 articles. For such high inclusion rates, a question can be raised whether the authors can provide quality analysis of the primary sources. the bold text in figure 11 indicates that the review has a quality score over 3, for topics that have more articles, such as Precision Agriculture and Forestry, there are relatively few high-quality reviews. However, high and low-quality SLRs are present across various topics, especially low-quality SLRs seem to be the norm rather than an exception. Finally, the italic text in figure 11 indicates whether systematic data extraction tables are presented in the review, as supplementary data, appendices, or in-text. Only 27% of included articles provide this clear overview of selected studies and/or preferred processing methods.

B. RQ1. WHAT IS THE QUALITY OF SYSTEMATIC

LITERATURE REVIEWS CONDUCTED IN THE UAV DOMAIN? The histogram in figure 5 shows that even though the average quality of the included reviews is 2.13, there is a high peak on the lower side of the quality scores. Not a single study acquired the highest rating of 4.

The average results from the quality assessment are shown over time in figure 6. A slight growth in quality is observed over time. However, the average score of 2.13 (out of 4) is not particularly high.

The lowest scoring quality criteria was QA3 with an average of 0.17 out of 1 (yellow in figure 6. QA3 asked whether the review performed a quality control check on their primary studies, which often is not executed in UAV systematic reviews. Additionally, there were articles that said to have performed a quality assessment but did not report the method of this assessment in a clear manner. For instance, the work of [3] presents four quality assessment questions, but no rating scheme for these questions, nor the results of the quality assessment.

The three articles that reported on a full quality assessment, were [23], [24], [25]. In table 2, the quality assessment types of these three articles are described. In [23], most of their included studies score a 'good', which is the middle score, with very few 'average' or 'state-of-the-art' ratings. Reference [24] only reported on a single quality criterion, although they rated their primary research on four criteria.



FIGURE 5. Histogram on the overall quality of included SLRs.



FIGURE 6. Average QA score grouped by year and averaged over the score. Individual articles can have higher scores than shown here.

This single criterion is reported as a number between zero and six, which means little without the context, which is not given. Also missing is an overall score on the quality of their primary research. The 20 quality questions and the reporting on the quality assessment performed in [25] is the most useful, presenting an average score of 64% out of 100%. Furthermore, they indicate where primary research is lacking; reporting on soil types, cultivar type, weather conditions, number of taken images, time of day, velocity, and flight duration. They also separated their results according to remote-sensing versus plant-protection-based studies, these two fields indicate similar quality results, and show high overlap in the above-mentioned missing values.

By examining the other low QA scores in this study, QA2 also does not show particularly high ratings either (green in figure 6). This criterion covers whether it is likely that the review has covered all relevant literature. This is measured by looking at the number and diversity of databases that have been searched. Many only include two or three databases (n=30), whilst some searched in only a single database or did not report their used databases at all (n=21).

TABLE 2. Reviews with full quality assessment.

Topic	Assessment type
Water	On a three-point scale: rated accu-
Management	racy & generalization of ML models
	of primary studies
Drone Technolo-	Rationale for study, adequate de-
gies	scription, clear findings, limita-
	tions, adapted from [10]
Precision	20 items on technical and environ-
Agriculture	mental factors relevant to UAV stud-
	ies, adapted from sports and medi-
	cal research
	Topic Water Management Drone Technolo- gies Precision Agriculture

TABLE 3. Reported hardware items with varying levels of detail.

Level of Detail	No.	reported
	items	
Low: Platform type (eg. 'Fixed-wing')	1035	
High: Platform brand & model (eg. 'senseFly	750	
eBee')		
Low: Payload type (eg. 'Multispectral cam-	1151	
era')		
High: Payload brand & model (eg. 'Tetracam	188	
ADC-lite)		

C. RQ2. WHAT ARE THE POPULAR UAV PLATFORMS, PAYLOADS, AND SOFTWARE IN THE UAV DOMAIN?

Out of the 21 SLRs that performed a detailed data extraction (italic in figure 11), 14 provided detailed information on hardware. For hardware, this resulted in 1208 primary studies. From these studies, different levels of detail were reported, which varies from study to study. Table 3 shows that more studies report on a low level of detail: 'quadcopter' versus a high level of detail: 'DJI Phantom 4 Pro', for payload extraction this difference is even larger.

1) UAV PLATFORMS

In figure 7, the growth of UAV research is clearly visible, as the image is based on primary research. Articles after 2020 have been excluded from this image, as only a few SLRs included articles after 2020. The colour indicates which UAV type was used in the primary research. The availability of multi-rotor UAVs has been the main cause of growth in UAV research since 2012 (pink/purple in figure 7). Most of the used multi-rotors have a quadcopter design. The fixed-wing design has also been used in a significant portion of the published work. Helicopters have almost disappeared completely, although they were sometimes used before 2016. Some prior SLRs may have included balloons as a UAV. However, in recent years, the balloon is not seen as a UAV anymore due to the growth and dominance of the multi-rotor. Not reported in the studies is the hybrid-wing, sometimes called VTOL fixed-wing (Vertical Take Off and Landing). This UAV design contains wings for long-distance flight, and capabilities for hover and horizontal take-off and landing similar to a quadrotor [26]. It could be that this design is too recent or experimental for inclusion in the studies.

DJI is by far the most used brand in primary studies (figure 8). No doubt that the advanced autopilot, reliable operation, stabilized cameras, and their ready-to-fly ecosystem



FIGURE 7. Platform occurrence by year and type. Data is based on the release year of the primary source and the UAV type. SLRs not indicating details on UAV model or type have been aggregated to multi-rotor.

influence the popularity greatly. The 'Others' category is also quite large, indicating a high diversity in UAV platforms being used. Furthermore, senseFly is also popular. The senseFly company exclusively develops fixed-wing type UAVs. This type is a smaller section of all used UAV platforms (see figure 7). The focus from senseFly on long flight times and precision mapping no doubt has an influence on the popularity in academia. Other popular brands shown in figure 8, such as HiSystems, 3DR, and AscTec have all gone out of business or merged into larger companies, such as Intel or Leica [27]. However, using custom UAVs is also a popular alternative to the off-the-shelve options ('Custom' in figure 8). Perhaps the open hardware and software history of UAVs [14], [15] plays a role in the ongoing popularity of the Do-It-Yourself UAV.

There is a dependence on only a few manufacturers for supplying the flying platforms that are used to perform research. The further implication of this dependence does vary in scale but deserves further attention.

2) UAV PAYLOADS

The most common sensor types over time are shown in figure 9, the digital camera is the most popular. Multispectral and hyperspectral cameras are significantly less used but have seen regular use since the earlier beginnings. Thermal and LiDAR sensors see the least use, although in 2019 and 2020 LiDAR has become more prevalent, perhaps indicating an increase in popularity.

The digital camera is often good enough for many research fields where photogrammetry and/or 3D information suffices to provide insight, such as structural engineering, waste management, and traffic management [28], [29], [30]. Also, the simplicity of interpretation and use could explain the popularity. Additionally, the digital camera is also often built into many UAV platforms, such as the DJI Mavic and Phantom line-ups.



FIGURE 8. Popularity of UAV platform brands, based on the occurrence in primary articles, sorted by top 10 occurring brands. 'Others' is the combination of all other brands not included in this top 10, consisting of 118 different UAV Brands.



FIGURE 9. Payload occurrence by year and type.

However, there is a high diversity in sensor use, apart from the most popular ones shown here, mainly used for mapping. The SLRs reported 46 different sensor types. This indicates high flexibility to the UAV, which is creatively used in primary research: there are studies adding particle sensors, gas sensors, microphones, networking equipment, and agricultural sprayers to UAVs.

The popularity of brands for UAV payloads (figure 10) is less skewed to a single manufacturer than the platforms are. Digital camera manufacturers Canon, Sony, Panasonic, and GoPro all have their cameras used in UAVs. Some primary research has modified these cameras to a wider range of light, by replacing the internal infrared filter, making them more akin to a multispectral sensor. Multispectral sensor manufacturers Tetracam and MicaSense also see a significant share of the popularity. DJI is also popular, likely due to the integrated solutions they provide. FLIR focuses on Thermal sensors, Headwall on the hyperspectral side of sensing, and Vaisala on gas and particle sensing equipment. A manufacturer often focuses on a specific sensing niche: hyperspectral sensing for Headwall, Thermal for FLIR and Infratec, Multispectral for Tetracam, etc. The adoption of



FIGURE 10. Popularity of payload brands, based on the occurrence in primary articles, sorted by top 10 occurring brands. 'Others' is the combination of all other brands not included in this top 10, consisting of 27 different UAV Brands.

their sensors within UAV usage is often only a part of their businesses. The outlier here is DJI, which develops a wider range of sensor types specifically for its UAV platforms. The payload brand popularity plot (figure 10) only shows the 188 primary articles on which data was extracted on brand and model, so might be less representative of popularity in the real world.

3) UAV SOFTWARE

Out of the 21 SLRs performing detailed data extraction, 11 SLRs performed a detailed software data extraction. A study using UAVs is built on various software: a mission planner app for the planning stage; flight control software and perhaps experimental obstacle avoidance algorithms in the flight stage; and the analysis stage might make use of Deep Learning, photogrammetry, or a simple linear regression. Therefore, an important distinction must be made in UAV software between software packages (e.g. QGIS 3.11, Agisoft Metashape or PyTorch) and data processing approaches (e.g. spatial overlap and joins, Structure from Motion or Yolov5 object detection). The exact software package that was used in the primary source is often not reported: only three out of the 11 SLRs with a software extraction [1], [31], [32] reported on the software packages. Additionally, confusion exists: [22] provided a software package for a data processing approach. From these four SLRs, 32 different software packages, for 10 different usages have been reported, the list in table 4 is by no means a comprehensive list of all available packages, but a broad categorization based on these four SLRs. Additionally, these four SLRs all report on some form of UAV usage for image acquisition, creating a bias toward post-processing software packages, and missing real-time packages such as ROS and its library of real-time robotics software [33].

Additionally, table 4 does not report the *data processing approaches* (such as quantitative inversion modeling, Deep Learning or Structure from Motion). Nine out of 11 SLRs reported on data processing approaches. Only [31] reported on both the data processing approach and the software

23228

packages used in the primary source. However, all of the nine SLRs that systematically extracted the approach have only reported on a single model in the approach. Three SLRs [22], [34], [35] also included the accuracy score of the corresponding model, such as r^2 for a quantitative inversion model [34]. Problematically, a processing approach is more than just a single model and its accuracy score: a processing approach can be performed in multiple software packages, and/or combine multiple approaches and packages. For example, a primary study included in [35], shows a biomass estimation approach for forestry [36]. This primary study reports on five steps: 1. a UAV was flown to acquire a set of images of a forest (not mentioned which flight planning approach or package has been used), 2. these images were processed into a 3D model (using Bentley ContextCapture), 3. a canopy height model (CHM) was calculated using Terrasolid, 4. the CHM was imported in eCognition to perform the tree-segmentation algorithm, and finally 5. a linear regression model was fitted to the data to estimate the biomass for each tree in the dataset (not mentioned which software package). The example shows how complex a single approach can be. Systematically extracting a complete approach is therefore quite difficult, which is perhaps the reason why none of the included SLRs attempted it.

D. RQ3. WHICH LIMITATIONS FOR USING UAVS ARE IDENTIFIED IN THE SLRS?

Different research fields have adopted UAV technology as part of their toolkit. These adoptions range from fields that are already deploying UAVs daily [29], to others that are just getting started [8]. Cross-pollination between these fields is seen as an important step in improving and adapting UAV research as a whole [4]. This section presents a shared set of limitations for deploying and researching UAVs in varying research fields. Four recurring limitations were identified from the articles with a quality score over 3 (bold in figure 11, n=16): regulatory, technical, social, and research limitations (see table 5).

Regulations are lagging behind the technology [30], [37], [43], resulting in vague criteria, such as noise levels [38]. Or regulations not being adaptive enough for disaster response activities [39]. Additionally, companies can take advantage by implementing unregulated technologies, with the prospect of a (temporary) monopoly [44], resulting in unequal access to this technology and/or its benefits [25]. Finally, regulations can be too limiting, many regulatory frameworks do not allow for beyond-visual-line-ofsight (BVLOS) flights [38], which are required for future autonomous applications.

Technical limitations differ for various fields: each field has its own demands. Fields using UAVs for sensing identify that the top-down imagery obfuscates details in plants [31], requiring more advanced flight paths or levels of autonomy [24]. Additional image processing limitations exist in low-quality of photogrammetry [29]. Correctly

Flight control DJ GO [32] Flight app for DJI UAVs Flight planning MAVinci [32] ArduPilot open source project for autopiloting UAVs Pix4DCapture [1], [31], [32] ArduPilot open source project for autopiloting UAVs DJI Flight Planner [32] Image Editors Adobe Photoshop [32] Image Rectification Rese App PARGE [32] Processing PixelWrench 2 [1], [32] Processing Vorkswell Core Player [32] Photogrammetry suites CSIRO Scylarrus [32] Photogrammetry suites AudPano Giga [32] Photogrammetry suites AudPano Giga [32] Photogrammetry suites AudPano Giga [32] Point Cloud processing CoudCompare [32] Proint Cloud processing CloudCompare [32] Point Cloud processing CloudCompare [32] DronoLoploy [31] Remote sensing + Photogrammetry suite Geo-Information package BAS SOCET GXP [1] Point Cloud processing CloudCompa	Туре	Software name	SLR	Notes
Flight planningMAVinci[32]Mission Planner[1]ArduPilot open source project for autopiloting UAVsPix4DCapture[1], [31], [32]DroneDeploy[1], [31]DII Flight Planner[32]Image EditorsAdobe PhotoshopImage RectificationRese Apps PARGEMultispectralpre- processingHyperspectralpre- processingThermal pre-processingWorkswell Core PlayerPhotogrammetry suites[32]AutoPano Giga[32]Aligning images for panoramic photosPhotogrammetry suites[32]AutoPano Giga[31]Prot Cloud processingCloudComparePoint Cloud processingCloudComparePoint Cloud processingCloudComparePoint Cloud processingCloudCompareGeo-Information packageBAS SOCET GXPPoint Cloud processingCloudCompareCloudCompare[32]ERDAS Imagine[32]Rese Cliss[31]Geo-Information packageBAS SOCET GXPProgramming languages for Data analysis, using Machine or Deep Learning methodsPythonRes[32]Programming languages for Data analysis, using Machine or Deep Learning methodsPythonRes[32]PointodsRRes[32]	Flight control	DJI GO	[32]	Flight app for DJI UAVs
Mission Planner[1]ArduPilot open source project for autopiloting UAVsPix4DCapture DroneDeploy[1], [31], [32] [11], [31]autopiloting UAVsImage EditorsAdobe Photoshop GIMP2[32]Data labeling Open source photoshop programImage RectificationRese Apps PARGE[32]Data labeling Open source photoshop programImage RectificationRese Apps PARGE[32]Specific to Tetracam sensorsProcessing Hyperspectral processingWorkswell Core Player[32]Specific to Workswell sensorsHutoPano Giga[32]Aligning images for panoramic photosAligning images for panoramic photosPhotogrammetry suitesAutoPano Giga[31]Aligning [31]Leica Photoparametry Suite AgtSoft Metashape[32]Aligning images for panoramic photosPoint Cloud processingCloudCompare LASTools[32]Remote sensing + Photogrammetry suiteGeo-Information packageBAS SOCET GXP[11, [32] [32]Remote sensing focusedESRI ArcGIS CGSIS[11, [32] [32][32] [32]Remote sensing focusedProgramming and pagages for Data analysis, using Machine or Deep Learning methodsPython[31]Programming analysis, using Machine or Deep Learning methodsPython[31]	Flight planning	MAVinci	[32]	
Pix4DCapture DroneDeploy DIF Flight Planner[1], [31], [32] [1], [31]It T SImage EditorsAdobe Photoshop GIMP2[32]Data labeling Open source photoshop programImage RectificationRese Apps PARGE[32]Data labeling Open source photoshop programImage RectificationRese Apps PARGE[32]Specific to Tetracam sensorsProcessing Thermal pre-processingWorkswell Core Player[32]Specific to Workswell sensorsPhotogrammetry suitesAutoPano Giga[32]Aligning images for panoramic photosPhotogrammetry suitesLeica Photogrammetry Suite AgiSoft Photoscan Professional AgiSoft Metashape[32]Aligning images for panoramic photosPoint Cloud processing DroneDeploy Pix4DCloudCompare [32][32]Remote sensing + Photogrammetry suiteGeo-Information packageBAS SOCET GXP[1][31]Remote sensing focusedGeo-Information packageBAS SOCET GXP[1][32]Programming languages for Data analysis, using Machine or Deep Learning methodsPython[31]Programming languages for Data analysis, using Machine or Deep Learning methodsPython[31]Programming languages for Data analysis, using Machine or Deep LearningPython[31]Programming languages for Data analysis, using Machine or Deep LearningPython[31]Programming languages for Data analysis, using Machine or Deep LearningPython[32]Programming languages for Data analysis, usingPython<		Mission Planner	[1]	ArduPilot open source project for autopiloting UAVs
DroneDeploy DII Flight Planner[1], [31]Image EditorsAdobe Photoshop[32]Image RectificationRese Apps PARGE[32]Image RectificationPre- ProcessingPixelWrench 2[1], [32]Multispectral processingpre- ProcessingCSIRO Scylarrus[32]Thermal pre-processing Photogrammetry suitesWorkswell Core Player[32]Specific to Workswell sensorsAutoPano Giga[32]Aligning images for panoramic photosPhotogrammetry suitesLeica Photogrammetry Suite[32]Aligning images for panoramic photosMCMAC Sigoft Photoscan Professional AgiSoft Metashape DroneDeploy[31] [31]PixetDPoint Cloud processing Could compare LastodsCloudCompare [32][32]Point Cloud processing CloudCompare LASTools[32]Remote sensing + Photogrammetry suiteGeo-Information package BAS SOCET GXP[1], [32] [32]Remote sensing focusedProgramming languages for Data analysis, using Machine or Deep Learning methods[31]PixelDProgramming languages for Data analysis, using Phytion[31]PixelDProgramming languages for Data analysis, usingPhytion[31]Programming languages for Data analysis, usingPhytion[31]Programming languages for Data analysis, usingPhytion[31]Programming languages for Data analysis, usingPhytion[31]Programming languages for Data analysis, usingPhytion[31]P		Pix4DCapture	[1], [31], [32]	1 0
Dil Flight Planner[32]Image EditorsAdobe Photoshop[31]Open source photoshop programGIMP2Image RectificationRese Apps PARGE[32]Multispectralpre- processingPixelWrench 2[1], [32]Specific to Tetracam sensorsSpecific to Tetracam sensorsprocessingVorkswell Core Player[32]Photogrammetry suitesAutoPano Giga[32]Photogrammetry suitesAutoPano Giga[32]Photogrammetry suitesLeica Photogrammetry Suite AgiSoft Metashape[31] [31] [31]Point Cloud processingCloudCompare LASTools[32]Point Cloud processingCloudCompare LASTools[32]Point Cloud processingCloudCompare LASTools[32]Point Cloud processingCloudCompare LASTools[32]Geo-Information packageBAS SOCET GXP[1] [32]Point Cloud processingGISS[31] [32]Geo-Information packageBAS SOCET GXP[1] [32]Point Cloud processing[31], [32] [32]Geo-Information packageBAS SOCET GXP[1] [32]Programming languages for Data analysis, using[31] Python[31]Programming languages for Data analysis, usingPython[31] [31]Programming languages for Data analysis, usingPython[31]Programming languages for Data analysis, usingPython[31]Programming languages for Data analysis, usingPython[31]Programmi		DroneDeploy	[1], [31]	
Image Editors Adobe Photoshop [32] Data labeling Image Rectification Rese Apps PARGE [31] Open source photoshop program Multispectral pre- processing PixelWrench 2 [1], [32] Specific to Tetracam sensors Hyperspectral pre- processing CSIRO Scylarrus [32] Specific to Workswell sensors Photogrammetry suites AutoPano Giga [32] Aligning images for panoramic photos Leica Photogrammetry Suite [32] [31] AgiSoft Metashape [31] [31] DroneDeploy [31] [31] Point Cloud processing CloudCompare [32] Geo-Information package BAS SOCET GXP [1] eCognition Developer [32] [32] Harris GeoSpatial ENVI [32] Remote sensing focused ESRI ArecIIS [1], [32] [32] GaASS GIS [32] [32] Programming Ianguages [31] Proprocessing [31] [32] Point Cloud processing CloudCompare [32] LASTools [31] [32] Geo-Information package BAS SOCET GXP [1] RENDAS Imagine [32] [32] ESRI ArecIIS		DJI Flight Planner	[32]	
Image RectificationGIMP2[31] (32]Open source photoshop programImage RectificationPresoProcessing[32]Open source photoshop programMultispectralpre- processingPixelWrench 2[1], [32]Specific to Tetracam sensorsHyperspectralpre- processingCSIRO Scylarus[32]Specific to Workswell sensorsPhotogrammetry suitesAutoPano Giga[32]Aligning images for panoramic photosPhotogrammetry suitesAutoPano Giga[31][31]Leica Photogrammetry Suite AgiSoft Metoscan Professional AgiSoft Metoscan Professional Prix4D[31][31]DroneDeploy[31][31]Point Cloud processingCloudCompare LASTools[32]Remote sensing + Photogrammetry suiteGeo-Information packageBAS SOCET GXP[1], [32]Remote sensing focusedESRI ArcGIS 	Image Editors	Adobe Photoshop	[32]	Data labeling
Image RectificationRese Apps PARGE[32]IIIIMultispectral processing Hyperspectral processingpre- PixelWrench 2[1], [32]Specific to Tetracam sensorsHyperspectral processingpre- processingCSIRO Scylarrus[32]Specific to Workswell sensorsPhotogrammetry suitesAutoPano Giga[32]Aligning images for panoramic photosPhotogrammetry suitesAutoPano Giga[32]Aligning images for panoramic photosPhotogrammetry suitesAutoPano Giga[31]DroneDeploy Pix4D[31] UroneDeploy[31]Point Cloud processingCloudCompare LASTools[32]Point Cloud processingCloudCompare LASTools[32]Geo-Information packageBAS SOCET GXP[1]Remote sensing + Photogrammetry suite[32]Remote sensing focused[31]BAS SOLET GXP[1], [32] [32]Grant Sub Simaging CRAS GIS SAGA GIS Q-GIS[31]Programming languages for Data analysis, using Machine or Deep Learning methodsR[32]Programming languagesPython[31]Programming languagesRemote Python[31]Programming languagesR[32]Staff or Data analysis, using Machine or Deep Learning methodsR[32]	e	GIMP2	[31]	Open source photoshop program
Multispectral processingpre- processingPixelWrench 2[1], [32]Specific to Tetracam sensorsHyperspectral processingpre- processingCSIRO Scylarrus[32]Specific to Workswell sensorsPhotogrammetry suitesAutoPano Giga[32]Aligning images for panoramic photosPhotogrammetry suitesLeica Photogrammetry Suite AgiSoft Photoscan Professional AgiSoft Photoscan Professional AgiSoft Metashape[31] [31]Point Cloud processingCloudCompare LASTools[32]Point Cloud processingCloudCompare LASTools[32]Geo-Information packageBAS SOCET GXP eCognition Developer Harris GeoSpatial ENVI ESRI ArcGIS[1], [32] [32]Programming GRASS GIS SAGA GIS Q-GIS[31]Remote sensing focusedProgramming Machine or Deep Learning methodsR[32]Remote sensingPython[31]Remote sensing[32] [32]Remote sensing focused[32]	Image Rectification	Rese Apps PARGE	[32]	
Protection pre- processing CSIRO Scylarrus [32] Specific to Workswell sensors Photogrammetry suites AutoPano Giga [32] Aligning images for panoramic photos Photogrammetry suites AutoPano Giga [32] Aligning images for panoramic photos Leica Photogrammetry Suite AgiSoft Photoscan Professional AgiSoft Photoscan Professional AgiSoft Photoscan Professional AgiSoft Attabape [31] DroneDeploy [31] [32] Point Cloud processing CloudCompare LASTools [32] Geo-Information package BAS SOCET GXP [1] Remote sensing + Photogrammetry suite eCognition Developer [32] Remote sensing focused suite Geo-Information package BAS SOCET GXP [1] Remote sensing focused ESRI ArcGIS [1], [32] [32] Remote sensing focused GRASS GIS [32] [32] [32] [32] Programming languages Python [31] [32] [32] Programming languages Python [31] [32] [32] Machine or Deep Learning methods R [32] [32] [32]	Multispectral pre-	PixelWrench 2	[1], [32]	Specific to Tetracam sensors
processing Thermal pre-processingWorkswell Core Player[32]Specific to Workswell sensorsPhotogrammetry suitesAutoPano Giga[32]Aligning images for panoramic photosLeica Photogrammetry Suite AgiSoft Photoscan Professional AgiSoft Metashape[31][32]DroneDeploy[31][31]Pix4D[1], [32]MICMAC[22]Smart3DCapture[32]Cloud processingCloudCompareLASTools[31]Geo-Information packageBAS SOCET GXPHarris GeoSpatial ENVI[32]ERDAS Imagine[32]ESRI ArcGIS[31], [32]GRASS GIS[32]SAGA GIS[32]Q-GIS[1]Programming languages for Data analysis, using Machine or Deep Learning methodsPythonR[32]R[32]	Hyperspectral pre-	CSIRO Scylarrus	[32]	
Internal pre-processing Workswein Core Prayer [32] Specific is Workswein Sensors Photogrammetry suites AutoPano Giga [32] Aligning images for panoramic photos Leica Photogrammetry Suite [32] [31] Aligning images for panoramic photos AgiSoft Metashape [31] [31] proneDeploy [31] DroneDeploy [31] [32] [32] Point Cloud processing CloudCompare [32] [32] Geo-Information package BAS SOCET GXP [1] Remote sensing + Photogrammetry suite Harris GeoSpatial ENVI [32] gaite sensors ESRI ArcGIS [1], [32] [32] Remote sensing focused Programming languages for Deoploger [32] [32] Remote sensing focused Programming languages for Deoploger [32] [32] [32] Programming languages for panoramic Python [31] [32] Remote sensing focused [32] [32] [33]	Thermal are processing	Warkawall Care Playar	[20]	Specific to Workswall concern
Photogrammetry suitesAutoPano Giga[32]Aligning images for panoramic photosLeica Photogrammetry Suite AgiSoft Metashape[31][31]JoneDeploy[31][31]Pix4D[1], [32]MICMAC[22]Smart3DCapture[32]Point Cloud processingCloudCompare LASTools[32]Geo-Information packageBAS SOCET GXP[1]eCognition Developer Harris GeoSpatial ENVI[32]ESRI ArcGIS[1], [32]ESRI ArcGIS[1], [32]GrASS GIS Q-GIS[32]Programming languages for Data analysis, using Machine or Deep Learning methodsRR[32]R[32]R[32]R[31]R[32]R[31]R[32]R[33]R[33]	I hermai pre-processing	workswell Core Player		Specific to workswell sensors
Leica Photogrammetry Suite AgiSoft Photoscan Professional AgiSoft Photoscan Professional Photoscan Professional Photoscan Professional AgiSoft Photoscan Professional AgiSoft Photoscan Professional Photoscan Professional AgiSoft Photoscan Professional Photoscan Professional Photoscan Professional Photoscan Professional Photoscan	Photogrammetry suites	AutoPano Giga	[32]	photos
AgiSoft Photoscan Professional AgiSoft Metashape[1], [32] [31]DroneDeploy[31]Pix4D[1], [31], [32]MICMAC[22]Smart3DCapture[32]LASTools[31]Geo-Information packageBAS SOCET GXPERDAS Imagine[32]Harris GeoSpatial ENVI[32]ESRI ArcGIS[1], [32]ESRI ArcGIS[1], [32]GRASS GIS[32]SAGA GIS[32]Q-GIS[1]Programming languages for Data analysis, using Machine or Deep Learning methodsPythonR[32]R[32]R[32]R[31]R[31]		Leica Photogrammetry Suite	[32]	
AgiSoft Metashape DroneDeploy[31] [31]Pix4D[1], [31], [32]MiCMAC[22]Smart3DCapture[32]Point Cloud processingCloudCompare LASTools[31]Geo-Information packageBAS SOCET GXP[1]Remote sensing + Photogrammetry suiteeCognition Developer Harris GeoSpatial ENVI ERDAS Imagine[32]ERDAS Imagine GRASS GIS[32] [32]Remote sensing focusedProgramming languages for Data analysis, using Machine or Deep Learning methodsPython[31]Remote or Deep Learning methodsR[32]R[32][31]		AgiSoft Photoscan Professional	[1], [32]	
DroneDeploy PixAD[31] [1], [31], [32] [22] Smart3DCapture[32]Point Cloud processingCloudCompare LASTools[32]Geo-Information packageBAS SOCET GXP[1]Geo-Information packageBAS SOCET GXP[1]eCognition Developer Harris GeoSpatial ENVI ESRI ArcGIS[32]Remote sensing + Photogrammetry suitesuiteeCognition Developer GRASS GIS[32]garantic of the sensing hanguages for Data analysis, using Machine or Deep Learning methodsPythonR[32]R[32]R[32]R[32]R[32]R[31]		AgiSoft Metashape	[31]	
Pix4D[1], [31], [32]MICMAC[22]Smart3DCapture[32][20][32]LASTools[31]Geo-Information packageBAS SOCET GXPeCognition Developer[32]Harris GeoSpatial ENVI[32]ESRI ArcGIS[1], [32]ESRI ArcGIS[31], [32]GRASS GIS[32]SAGA GIS[32]Q-GIS[1]Programming languages methodsPythonR[32]R[32]R[32]R[31]		DroneDeploy	[31]	
MICMAC[22]Smart3DCapture[32]Point Cloud processingCloudCompareLASTools[31]Geo-Information packageBAS SOCET GXPeCognition Developer[32]Harris GeoSpatial ENVI[32]ERDAS Imagine[32]ESRI ArcGIS[1], [32]GRASS GIS[31]Or Data analysis, using methodsPythonRemote[31]Remote[32]Remote[32]Remote[32]Remote[32]Remote[32]Remote[32]Remote[32]SAGA GIS[32]Q-GIS[1]Programming languagesPython[31][31]R[32]		Pix4D	[1], [31], [32]	
Smart3DCapture[32]Point Cloud processingCloudCompare LASTools[31]Geo-Information packageBAS SOCET GXP[1]Remote sensing + Photogrammetry suiteeCognition Developer Harris GeoSpatial ENVI[32] [32]Remote sensing focusedESRI ArcGIS ESRI ArcGIS GRASS GIS GRASS GIS for Data analysis, using methods[31][32]Programming languages methodsPython[31]Remote sensing (31]Python[31]Remote sensing (32][32][32]Remote sensing focused[32]Remote sensing focused[31]SAGA GIS (32][32]R[31]Programming languages for Data analysis, using methodsRR[32]		MICMAC	[22]	
Point Cloud processingCloudCompare LASTools[32] [31]Geo-Information packageBAS SOCET GXP[1]Remote sensing + Photogrammetry suiteeCognition Developer Harris GeoSpatial ENVI[32] [32]Remote sensing focusedESRI ArcGIS[1], [32] [31], [32]Remote sensing focusedESRI ArcGIS[31], [32] [32]GRASS GIS [32][31]Programming languages for Data analysis, using methodsPython[31]Remote or Deep Learning methodsR[32]		Smart3DCapture	[32]	
LASTools[31]Geo-Information packageBAS SOCET GXP[1]Remote sensing + Photogrammetry suiteeCognition Developer Harris GeoSpatial ENVI[32]Remote sensing focusedERDAS Imagine ESRI ArcGIS[31], [32]Remote sensing focusedGRASS GIS GRASS GIS GRASS GIS GOIS[31], [32]Remote sensing focusedProgramming languages methodsPython[31]Remote sensing[31][32]Remote sensing[32]Remote sensing focused[32]ESRI ArcMap GRASS GIS SAGA GIS Q-GIS[32]R[31]	Point Cloud processing	CloudCompare	[32]	
Geo-Information packageBAS SOCET GXP[1]Remote sensing + Photogrammetry suiteeCognition Developer Harris GeoSpatial ENVI[32] [32]Remote sensing focusedERDAS Imagine ESRI ArcGIS[1], [32] [31], [32]Remote sensing focusedGRASS GIS GRASS GIS Q-GIS[31], [32] [32]Remote sensing focusedProgramming languages methodsPython[31]Remote sensing[31][32]Remote sensing focused[32]		LASTools	[31]	
eCognition Developer Harris GeoSpatial ENVI[32] [32] [32] [32] [32] [32] [32] [31], [32] [31], [32] [33], [32] [32] [33] [33] [32] [32] [32] [32] [32] [32] [32] [33] [32] [33] [32] [31]Remote sensing focusedProgramming languages for Data analysis, using methodsPython[31] [31]Remote sensing focused[32] [32] [32]	Geo-Information package	BAS SOCET GXP	[1]	Remote sensing + Photogrammetry suite
Harris GeoSpatial ENVI[32]ERDAS Imagine[32]ERDAS Imagine[32]ESRI ArcGIS[1], [32]ESRI ArcMap[31], [32]GRASS GIS[32]Q-GIS[1]Programming languagesPythonfor Data analysis, using methodsRR[32]		eCognition Developer	[32]	
ERDAS Imagine ESRI ArcGIS ESRI ArcGIS GRASS GIS Q-GIS[32] [1], [32] [31], [32] [32] [32] [32] [32] [32]Remote sensing focusedProgramming languages for Data analysis, using methodsPython[31] [31][31]R[32] [32][31][31]		Harris GeoSpatial ENVI	[32]	
ESRI ArcGIS[1], [32]ESRI ArcMap[31], [32]GRASS GIS[32]GRASS GIS[32]Q-GIS[1]Programming languagesPythonfor Data analysis, using Machine or Deep Learning methodsRR[32]		ERDAS Imagine	[32]	Remote sensing focused
ESRI ArcMap GRASS GIS SAGA GIS Q-GIS[31], [32] [32] [32] [32] [32] [32] [32]Programming languages for Data analysis, using Machine or Deep Learning methodsPython[31]R[32]		ESRI ArcGIS	[1], [32]	C C
GRASS GIS SAGA GIS Q-GIS [32] Programming languages for Data analysis, using Machine or Deep Learning methods Python R [32]		ESRI ArcMap	[31], [32]	
SAGA GIS Q-GIS [32] [1] Programming languages for Data analysis, using Machine or Deep Learning methods Python R [32]		GRASS GIS	[32]	
Q-GIS [1] Programming languages for Data analysis, using Machine or Deep Learning methods Python [31] R [32]		SAGA GIS	[32]	
Programming languages for Data analysis, using Machine or Deep Learning methods Python [31] R [32]		Q-GIS		
for Data analysis, using Machine or Deep Learning methods R [32]	Programming languages	Python	[31]	
Machine or Deep Learning methods R [32]	for Data analysis, using			
methods R [32]	Machine or Deep Learning			
R [32]	methods			
		R	[32]	

TABLE 4. Software packages reported in the included SLRs.

timing flights accounting for solar intensity, cloud cover and operational times is also a problem identified in research [25]. For other fields, the speed of communication between UAV operators and taking action can be too slow [3]. The need to acquire data and also process this data slows down the real-time decision-making, required in emergency response and healthcare [3], [42]. However, almost all fields identify the short battery life [8], [25], [40] as a problem. Larger areas that need to be covered require multiple flights, during which the weather and environment could change. Additionally, a high level of autonomy for UAVs is required for many advanced UAV usages, such as parcel delivery [24], [42].

Social factors are similar across various research fields, and include A negative public perception of UAVs due to propeller noise [38] and the military history [8]. UAV platforms are inherently dangerous devices, with multiple propellers being powered by powerful motors, and when one cuts out, the whole system falls to the ground. This means that the safety of operation towards humans, animals, buildings, and other vehicles always needs to be taken into account,

VOLUME 12, 2024

limiting adoption when this safety cannot be ensured [3], [24], [30].

Research limitations are also prominent, the SLRs are critical of the primary sources [25], [29]. The primary research covered in [31] on Deep Learning (DL) application to UAV imagery in forests uncovered that only a small number of primary studies used Deep Learning to a high standard. This high standard entailed the complexity of the Deep Learning task: single tree classification in a monoculture forest is not very complex, and segmenting treetops in a mountainous deciduous forest is a more complex task. Most studies only solved the simple task [31]. Another observation that was made concerned the testing-training split. A large number of their primary studies made at least one mistake with testing-training splitting. One example is splitting the data late in the analysis, resulting in a model that already has seen the test examples. Another example is splitting the data by randomly selecting pixels, this should be done by geographically splitting the data based on regions, preferably a different flight or area entirely [31].

Limitation	Details	SLR source	
Regulatory	Lagging	[25], [30], [37]	
	Vague	[38]	
	Too limiting	[25], [30], [39]	
Technical	Battery life	[8], [25], [40]	
	Level of autonomy	[24]	
	Obfuscating observations	[31]	
	Communication speed	[3]	
	Quality of photogramme-	[29]	
	try		
	Timing of flights	[25]	
Social	Public perception	[8], [38]	
	Safety of operation	[3], [24], [30]	
Research	Low standards	[31]	
	Lack of validation	[3], [24], [41]	
	No real-world tests	[3], [41], [42]	
	Underreporting	[25], [29], [30]	
	No cost-benefit	[29], [42]	

 TABLE 5. Recurring limitations indicated in UAV SLRs.

Furthermore, studies lack a validation step in their research [3], [24], [41]. This validation can be a field test, comparing versus other state-of-the-art, or ground-truth validation. Similarly, reviews also state that technical solutions are not tested in practice or real-life tests [3], [41], [42]. This can be an outdoor test and validation of medicine delivery or a multi-fire extinguishing task. The potentials of UAVs are not benchmarked against the default use, such as blood delivery by road and maternal healthcare provision by people [8]. Additionally, studies often underreport aspects of their research such as flight information, environmental factors [25], [29], [30] and which software was used [35]. Finally, a cost-benefit analysis or economic perspective to the study is often not executed [29], [42] even though the primary reasons to use a UAV are their accessibility and low cost [40].

V. DISCUSSION

A. UAV RESEARCH QUALITY

In the past decade, diverse studies identified, summarized and assessed the role that UAVs have played, and are playing in their research field. Despite the success of the technology itself, the quality of SLRs assessing the technology is lower than in other research fields.

The tertiary study on Machine Learning for Software Engineering [12], shows an average quality score around 3, rated on the same criteria as this TLR. Whilst the other UAV-related TLR scores a 0.3 [13], although missing one of the four DARE criteria in their assessment. Even if this fourth criterion would be a perfect score of 1, it would make it at best a 1.3 out of 4. This TLR presents an average quality score of 2.13, which is slightly higher than the TLR in [13]. Similar to findings in other TLRs [9], [13]: we found that SLRs are not performing a quality assessment on the primary research (see figure 6). Due to this missing information, little can be concluded about the overall quality of primary research in UAV studies.

However, table 5 indicates that SLRs are not satisfied with the primary research. Methodological errors were identified,

23230

validation steps were missing, no comparison versus other methods was performed, relevant variables were not reported, and cost-benefit analyses were missing. This is a grim reflection on the quality of UAV research. However, the trend of quality in reviews is slightly upward over time (6), and much improvement is to be gained if SLRs are going to perform quality assessment [9], [12].

In the supplement of this TLR, the QA questions and rating schemes from [23], [24], and [43] are provided (See supplement III. quality assessment: *UAVQualityAssessmentExamples.xlsx*). Additionally, the strict criterion that [31] presented in the study on deep learning can also be found in this supplement. These examples could be directly used or adapted to suit the topic of the SLR. The reader is encouraged to take a look at this supplement for identifying important features in a UAV study, even if they are not planning an SLR.

B. REPORTING UAV RESEARCH

Underreporting of variables has been identified as a problem in primary research dissemination by SLRs [25], [29], [30]. The underreporting of variables is also identified in this TLR, only 21 out of 73 studies performed a systematic data extraction. Perhaps the underreporting in primary research is a cause for secondary studies not to perform a data extraction: there is no data to extract. As elaborated in section IV-D (UAV Software), this problem of underreporting software packages and software processing approaches is even more profound. As indicated in this section, covering a UAV study systematically is a difficult problem: there are many variables that influence the final result of the study. To reckon with this problem, UAV primary research needs to report more details, and secondary research needs to align these details. This could finally open up the comparative analysis between UAV studies that has been missing since the dawn of this research field [15].

Inspired by the questions provided by [25], and recurring reported values in the secondary studies, detailed tables are provided (See supplement IV. primary research: *primary.xlsx*. An overview of this supplement is given in table 6. Additionally, reviews identify the future direction toward Deep Learning methods. An additional table provides Deep Learning-related aspects, which must be taken into account for UAV studies, inspired by the critical assessment from [31]. The variety of UAV studies should be embraced, these tables provide a basis and should be adapted to fit the needs of a UAV application and research field, such as adapting to UAV swarm studies, the combination of various sensing payloads, or autonomous parcel delivery experiments.

For future SLRs, empty data-extraction tables are provided (See supplement V. secondary: *secondary.xlsx*). An overview of the contents of these tables is given in table 11. A review should at least cover metadata on the journal, year, title and perform a quality assessment. In addition, information on the UAV experiment, analysis method, UAV platform, and payload also ought to be provided.

TABLE 6. Overview of parameters to report on in a UAV primary study.

Category	Parameters	Example variables	
Experiment	Flight	Date & time; Duration; Height;	
		Velocity; No. Images	
	UAV platform	Type/classification; Model	
		name; Takeoff weight; Cost	
Sensing	Sensor	Type/classification; Model	
		name; No. / type of bands;	
		Weight (g); Viewing geometry	
	Calibration	Procedure; Uncertainty value;	
		Reflectance calculation method	
Environment	Generic	Date & time Weather condi-	
		tions; Season; Wind speed	
	Agricultural	Soil type/conditions; Cultivar;	
		Phenology; Growth stage	
	Forestry	Forest type; Forest species; Soil	
		type	
Processing	Photogrammetry	Type; Settings; Software pack-	
		age; GSD; Overlap; Pixel size;	
		Accuracy report; No. Images	
	Data analysis	Problem description; Model	
		type/classification; Accuracy	
		type; Accuracy value	
	Deep Learning	Model name/architecture; Val-	
		idation set type; Test set type;	
		Pretrained weights; Tuned hy-	
		perparameters; Training time;	
		Carbon emissions	

C. CONSOLIDATING UAV TECHNOLOGIES

The rise of UAV technology has come from various different technological developments in the past decades, such as the continuous shrinking of electronics, higher capacity batteries and improvements in compute per unit power, algorithms to balance unstable aircraft, faster networking, and much more.

Simply combining these technologies does not form a UAV, it is the seamless integration of these technologies that enables flexibility and low cost to such a wide audience. Additionally, this seamless integration also includes regulatory and social aspects, such as reliability of flight (it does not stop or crash, goes to the home position with low battery) and conforms to regulations (weight limits, does not take-off in restricted areas unless a request is made to the authorities), whilst newer propellers are being developed to reduce the noise [45]. There is a complex system visible in a UAV platform: a system where state-ofthe-art hardware is combined to satisfy social norms through software development. This results in a platform that is difficult to get right, only a handful of manufacturers are able to do so and these are by a large margin the most popular manufacturers of UAV platforms included in this study (see figure 8). While others have gone under or have consolidated into larger companies. Switching manufacturers is not a simple decision. Vendor lock-in increases [46], [47]: proprietary communication protocols are already built-in to sensors and accessories such as radio controllers and payloads; costs of retraining for switching platforms; and the state-of-the-art increasing their lead on open source alternatives.

The UAV platforms have enabled various payloads to take the skies. From the total of 188 payloads that have been reported, there are 46 unique ones, although
 TABLE 7. Overview of data extraction parameters to report on in a UAV secondary study.

imaging sensors are the most frequent. This creativity in UAV applications will no doubt continue as access to flying platforms increases. Additionally, improvements in sensor technology such as higher resolutions, faster measurement intervals, and increased measurement sensitivity will have a direct impact on the capabilities of UAVs: more and more is possible.

Finally, the SLRs indicate a promising future using Deep Learning for UAV research [31], [48], [49], [50], [51]. These complex models offer various improvements in accuracy over previous approaches and are therefore seen as the future of the field. Realtime use-cases for UAV flight such as obstacle detection, accurate localization and mapping, and semantic task understanding are significantly improved by DL [4], [49]. Additionally, it is also the post-processing of the acquired data for disease mapping [25], [31], road segmentation [30], or crop yield prediction [49] that gets an accuracy boost from DL. However, these methods also present new problems, such as model biases, lack of labeled datasets , ethical risks, and environmental and financial costs [52].

There is an ongoing improvement in UAV technology: flight duration increases, sensors improve and analysis methods evolve [53]. Relevant technology may now become outdated the following year. However, mutual gain is easily achieved as long as a wide audience remains to have access to the state-of-the-art. This is done through open research: best practices are shared, hardware standards are developed, and software approaches are easily reproducible. Luckily, the open-source community has already developed many UAV software packages for wider adoption in various applications. Providing two-way interaction between academia and industry for improving the platform [54]. There are existing open source projects such as the PixHawk flight controllers [55], flight planning through Mission Planner [56], photogrammetry software such as Meshroom [57] and OpenDroneMap [58], analysis through QGIS [59] and GDAL [60], the ROS library [33] and more recently Deep Learning using the PyTorch [61], Keras [62] and Tensorflow [63] libraries.

VI. THREATS TO VALIDITY

The limitations of this TLR are identified using the Validity of Classifying Threats scheme from [19]. These threats are

TABLE 8. Study selection threats and mitigations this study.

Thursd	Midiantian	
Inreat	Mitigation	
Inadequacy of initial relevant	Systematic search string is pre-	
publications identification	sented in section 3.	
Limited number of journals and	Queried three databases and	
conferences	one indexing system used in	
	other computer science and	
	UAV research.	
Missing non-English papers	Purposefully excluded 7 arti-	
8 8 11	cles written in a different lan-	
	guage: 1% of all eligible pa-	
	pers.	
Paper not accessible in a digital	Only used digitally archived	
library	papers with access: 30 had no	
norary	access 10% of eligible papers	
Inefficient handling of dupli-	Duplicate was marked and	
cate articles	removed before the data-	
cate articles	extraction step duplicate	
	primary source articles are	
	marked in the date	
Inclusion/avaluation of grav lit	Gray literatura avaludad	
anotumo	for finding secondary	
erature	for finding secondary	
	studies, included grey	
	interature for searching	
	hardware specifications	
	from manufacturers websites.	
Insufficient study	Criteria defined in the review	
inclusion/exclusion criteria protocol, based on [9].		

TABLE 9. Data validity threats and mitigations in this study.

Thursd	Mitianting	
Inreat	Mitigation	
Small sample size or heteroge-	Included a high variety of re-	
neous primary studies	search fields and over 1000 pri-	
	mary articles identified.	
The chosen variables to be ex-	Discussed among authors, and	
tracted cannot answer the RQs	adapted from the included sec-	
	ondary reviews with data ex-	
	traction.	
Primary/secondary studies are	51 unique journals for the 73	
published in a limited number	SLRs.	
of venues		
The obtained dataset lacks rela-	Based on included SLRs with	
tionships	data extraction.	
Low validity of	Identified the highest scoring	
primary/secondary studies	SLRs for further analysis.	
Data extraction is biased	Data extraction was checked by	
	all authors. Although no kappa	
	statistics have been calculated.	
No statistical analysis of the	Not required to answer research	
dataset	questions.	
The selection of classification	Existing data-extraction	
schema is biased	schemas from the included	
	SLRs were used as a basis,	
	and updated continuously to fit	
	other SLRs that performed a	
	data-extraction.	
The interpretation of results is	Only articles with a quality	
not objective	score over 3 has been used to	
-	perform further analysis.	

subdivided into Study selection, Data validity, and Research validity. The threats are covered under their respective section, with the details and mitigations as tables.

A. STUDY SELECTION VALIDITY

The strategies used to search and select studies in this research are linked to the possibility of excluding relevant studies.

TABLE 10. Research Validity threats and mitigations in this study.

Threat	Mitigation
The lack of repeatability	Details of conducting the re-
	view has been shown in the
	methodology, and the extracted
	datasets have been made avail-
	able with this article.
An unfitting research method	The UAV field is growing sig-
has been selected	nificantly, a tertiary study is
	able to cover all these areas.
	Deviations in data extraction
	have been mentioned in the
	methodology.
Answering the RQs cannot ful-	The research questions have
fil the goal	been discussed by all authors
-	before conducting the review.
	These research questions have
	been used as structure through-
	out the methodology, results
	and conclusion to accomplish
	the research goal.
Lack of comparable studies	Related works includes one
	other UAV tertiary study and
	other tertiary studies in Ma-
	chine Learning and Software
	Engineering.
Researchers are not familiar	The related work section in-
with the research field	cludes both tertiary studies and
	UAV background.
Lack of generalizability	Findings comply with existing
	tertiary studies and included
	SLRs, the starting date of the
	search was set at 2012, which
	is confirmed by the time-series
	figures to be the start of the
	large growth.

The chosen year range for the search (2012-2022), preferred digital libraries, search terms used, and the selection criteria may have caused some studies to be missed. Expected is that broadening the query from *UAV* to also include variations, such as *RPAS*, *drone* and *UAS* would have increased the total search results. Additionally, the selection criteria for 'based on a defined search process' excluded many UAV review articles. However, this was done purposefully to only include SLRs, instead of any review. Finally, forward and backward snowballing could have been used to increase the number of SLR sources, however, the included 73 articles already covered a large variety of research fields, and there was a lot of overlap between the database results.

B. DATA VALIDITY

Acquiring different hardware, software, and themes from UAV studies requires many relevant variables to be checked and included. These variables are taken from the SLRs which performed a data extraction. This was the basis for a data-extraction schema. The used data-extraction schemas have been checked by the authors before starting the data-extraction process. However, many included articles did not provide a clearly structured table of their primary sources and the extracted variables, which are therefore not represented in the data extraction. For those that did present clearly structured tables, variation exists between

TABLE 11. Main findings.

Research question	Summary
RQ1. What is the quality of sys- tematic literature reviews con- ducted in the UAV domain?	The average score of the SLRs was 2.13 out of 4.The lowest-scoring criterion was related to whether a review performed a quality control step; many SLRs did not include this step.
RQ2. What are the popular UAV platforms, payloads, and software in the UAV domain?	High creativity observed in UAV primary research, driven by the flexibility of UAVs. Quadcopter is the most popular flying plat- form; fixed-wing and hybrid-wing aircraft gaining popularity. Depen- dence on a few manufacturers for UAV platforms in research, war- ranting further attention. Payloads are very diverse, but RGB sensors remain ever-popular. Limited re- porting on software-related aspects in SLRs; a distinction between soft- ware packages and data processing approaches is crucial.
RQ3. Which limitations for us- ing UAVs are identified in the SLRs?	See table 5.
Main RQ. How can future UAV studies be conducted to im- prove the quality of optimizing information sharing for practi- tioners and researchers?	Researchers: systematically address primary research, use the provided extraction tables and quality assessment questions as a starting point, and focus on developing open-source projects to maintain mutual benefits that made the field what it is today. Practioners: address the social and regulatory limitations of UAVs in a public manner, develop real- world use cases of UAVs and seek out market niches, commit to open-source projects and seek out collaboration.

study how details were presented. The schema was either adapted or the variable was adjusted: renaming variables to be more specific than originally presented: DJI P4 to DJI Phantom 4 Pro; Multirotor to Quadcopter (if the model was known). Finally, UAV technology moves rapidly, new hardware designs, features, and software updates make many older designs irrelevant and outdated. Therefore, using the information presented in the data-extraction tables to find the best hardware and/or software is discouraged.

C. RESEARCH VALIDITY

The research validity of this TLR is related to the generalizability of the results from our tertiary review to the population under study. Therefore, the representation of the selected articles should reflect the subject population. We tried to minimize this threat by searching different digital libraries and selecting all the results from the query, with a selected starting date of 2012, which is a meaningful starting date for the growth of UAV research, the oldest included article was published in 2016. The search, data extraction, and quality assessment have been performed by a single person which possibly introduced bias in these tasks, the resulting data extraction has been checked by all authors.

VII. CONCLUSION

This article presented a tertiary study on UAV applications. Based on 73 systematic literature reviews published between 2012 and 2022. This TLR aimed to assess the quality of SLRs, aggregate data across research fields, and provide broad guidelines to be applicable across the widespread UAV research community. The research questions supporting this aim are answered below.

RQ1. What is the quality of systematic literature reviews conducted in the UAV domain?

Quality assessment was performed on the included articles, based on criteria presented in [11]. This assessment revealed a mix of higher and lower-quality research. The average score of the SLRs was 2.13 out of 4. The lowest-scoring criteria was the criteria covering whether a review performed a quality control step in their SLR, many SLRs did not perform one. Therefore, four different example quality assessments from the SLRs are presented in supplement III, as a guideline or for direct adaptation in UAV studies.

RQ2. What are the popular UAV platforms, payloads, and software in the UAV domain? 21 out of the 73 articles performed detailed data extraction, covering hardware and software in UAV research. This extraction is performed to varying degrees of detail, where a low level of detail occurs more often. There is high creativity in UAV primary research, as the flexibility of the UAV enables many applications to take the sky. The quadcopter is by far the most popular flying platform, followed by the other multi-rotors, fixed-wing aircraft are also experiencing stable popularity, for certain applications hybrid-wing aircraft could also see growth in the coming years. There is a dependence on only a few manufacturers for supplying the flying platforms that are used to perform research. The further implications of this dependence vary in scale but are deserving of further attention. Payload-wise, if it is light enough to be mounted on a UAV, research is finding a way to make it fly. The different imaging sensors are the most popular, with the standard digital camera being the most popular. In the case of software, an important distinction needs to be made between the software package and the data processing approach. A software package is a set of libraries and executables to run a specific task. Whereas a data processing approach is the theoretical set of formulas and calculations processed in the software package. Both of these software-related topics are underreported in the SLRs. Therefore, little can be concluded about the popularity of specific packages or data processing approaches. Broad categories of used software packages fall under flight control, flight planning, image editing, sensor pre-processing, photogrammetry, geoinformation, and programming languages for analysis.

RQ3. Which limitations for using UAVs are identified in the SLRs?

The limitations of UAVs are social, technical, regulatory, and research-based. Where social limitations can be the negative perception of UAVs and the noise a UAV makes. The regulatory limitations are the lack of regulations and



FIGURE 11. Visual overview of included SLR articles. Different colours correspond to different topics in UAV-related research. The italic article shorthands are the reviews which provide detailed information on hardware or software used in the primary studies (n=20). The bold text are rated a quality score of 3 or higher (n=16). The left figure shows the start and end date of the search query of the SLR. And the right figure are the number of included articles that the SLR identified.

their clarity, or too stringent regulation halting UAV adoption in that jurisdiction. Technical limitations include limited flight time, payload weight, top or oblique views not showing enough, photogrammetry not resulting in accurate

representations, and the complexity of operating a UAV to a high standard. Finally, the SLRs indicate that primary studies are missing various important aspects: low-quality writing, the lack of validation, underreporting variables, and little to no benchmarking being executed, to compare UAV approaches versus other approaches.

How can future UAV studies be conducted to improve the quality of optimizing information sharing for practitioners and researchers?

Before UAV research can reach maturity, it needs to recon with its pubescent years. The first years of explorations and boundless possibilities are coming to an end, and UAVs will be mainstream technologies in the not-too-distant future. However, it is the improvement of research quality, technology, clear regulations, and social acceptance that are the roadblocks in getting UAVs of age. Therefore, a conclusion to the main research question is divided into advice for researchers and practitioners:

Recommendations for Researchers

Secondary research could improve significantly by systematically assessing primary research. Quality of research is not particularly high, in both primary and secondary research. Additionally, systematic data extraction should increase in detail, these details are what could enable larger comparative analysis, which is currently missing in UAV research at large. Provided in the supplement of this review are quality control questions, and data extraction tables, which are adapted from the included reviews. These should serve as a guideline for future research. The reader is therefore encouraged to take a look at these tables, to understand what ought to be reported in primary and secondary UAV research. Additionally, research fields should continue to devise and improve classification schemes and preferred reporting items for UAV research, improving adoption and replicability, whilst the technology continues to permeate across academia. Finally, UAV technology keeps rapidly improving, open source hardware and software should continue to be developed in research, ensuring generalization, the adoption of research, and mutual benefits of the rapid pace, whilst maintaining some independence from monopolizing companies.

Recommendations for Practitioners

The UAV practitioner is a concept that will grow in the coming years, as the adoption of UAVs will spread from research to industry, and more and more applications will see the rise of the flying platform. There is a variety of existing, viable UAV applications that do not cross paths with academia, such as professional photography and filming, drone racing, and a real-time eye-in-the-sky for police forces. Increasing collaboration with academia could provide fruitful in addressing the social and regulatory limitations of UAVs, through advocacy of more lenient and clear regulations and showing societal relevance. Additionally, regulations can also cover manufacturers to develop safe systems, by taking advantage of the intelligence built into the platform. Examples can be required obstacle avoidance,

0, 1	X 7	NZ	
Study	Venue	Year	QA
			Score
[23]	Sensors	2019	3.5
[31]	Remote Sensing	2021	3.5
[38]	International Journal of En-	2021	3.5
	vironmental Research and		
	Public Health		
[25]	Agronomy	2022	35
[2]	Forests	2022	3.5
[3]	Clobal baalth anidamalagy	2022	2.5
[42]	Global health, epidemology	2018	3
FO 41	and genomics	2010	2
[24]	Future Generation	2019	3
	Computer Systems		
[8]	Medicine	2020	3
[35]	Remote Sensing	2020	3
[40]	IEEE Access	2021	3
[39]	IEEE Access	2022	3
[30]	Remote Sensing	2022	3
[41]	Science & Justice	2022	3
[29]	Waste Management	2022	3
[51]	Remote Sensing	2022	3
[64]	Concurrency and Computa-	2022	3
[0,1]	tion - Practice & Experience		0
[65]	Physical Communication	2022	3
[05]	Dropas	2022	25
[00]	Animala	2019	2.5
[3]	Annais	2020	2.5
[0/]		2021	2.5
[68]	Computer Communications	2021	2.5
[7]	Forests	2021	2.5
[69]	Remote Sensing	2021	2.5
[70]	Remote Sensing of the Envi-	2021	2.5
	ronment		
[71]	Drones	2021	2.5
[49]	Journal of Robotics and	2022	2.5
	Control		
[72]	Engineering, Construction	2022	2.5
	and Architectural		
	Management		
[73]	Computers and Electronics	2022	25
[,5]	in Agriculture	2022	2.5
[74]	Sustainability	2022	25
[74]	Sustainability	2022	2.5
[75]	Interactive Learning Envi	2022	2.5
[/0]	Interactive Learning Envi-	2022	2.5
10 01	ronments	2016	2
[20]	Technology in Society	2016	2
[37]	Remote Sensing	2017	2
[77]	Construction Innovation	2018	2
[78]	IOP Earth and environmen-	2019	2
	tal science 2019		
[79]	International Journal of	2019	2
	Innovative Technology and		
	Exploring Engineering		
[80]	Technology in Society	2020	2
[22]	Consider Journal of Ecrest	2020	C
[22]	Research	2020	L

[43]	Applied Artificial	2020	2
	Intelligence		
[28]	Minerals	2020	2
[32]	Science of the Total Environ-	2020	2
	ment		
[81]	Precision Agriculture	2021	2
[48]	IEEE Access	2022	2
[82]	Remote Sensing	2022	2
	Applications: Society		
	and Environment		
[83]	Forests	2022	2
[84]	International Journal of Risk	2021	2
	Reduction		
[85]	Sugar Tech	2022	2
[18]	Sensors	2016	1.5
[34]	Frontiers in Plant Science	2017	1.5
[86]	European Journal of Remote	2018	1.5
[]	Sensing		
[87]	IGLC2018	2018	1.5
[1]	Drones	2018	1.5
[88]	Journal of Information Tech-	2019	1.5
	nology in Construction		
[89]	Environmental Conservation	2019	1.5
[90]	Remote Sensing of the Envi-	2019	1.5
_	ronment		
[91]	Remote Sensing	2020	1.5
[92]	Applied Sciences	2020	1.5
[2]	Transportation Research In-	2020	1.5
	terdisciplinary Perspectives		
[21]	Journal of Air Transport	2020	1.5
[93]	Smart and Sustainable Built	2021	1.5
	Environment		
[94]	Forests	2021	1.5
[95]	Sustainability	2021	1.5
[96]	Production Engineering	2022	1.5
[97]	Remote Sensing	2022	1.5
[98]	International Transactions in	2021	1.5
	Operational Research		
[99]	Journal of Intelligent and	2021	1.5
	Robotic Systems		
[100]	ACM SIGCOMM 2019	2019	1
[101]	ESREL 2019	2020	1
[50]	Remote Sensing	2020	1
[102]	Future Internet	2021	1
[103]	Remote Sensing	2021	1
[104]	Remote Sensing	2022	1
[105]	Arabian Journal for Science	2022	1
	and Engineering		

 TABLE 12. (Continued.) Included studies, ordered from high to low based on quality assessment score.

battery life notifications, and built-in geofencing before it can be sold. There is diverse research available on various UAV practices and analysis methodologies, practitioners can adapt these to develop real-world use cases of UAVs and seek out market niches. Traditionally, UAV platform and payload manufacturers have been providing steady improvements on the flying platforms such as reducing costs and increasing the capabilities, these improvements have enabled wider audiences to reap the benefits of the platform. To increase adoption and reduce technical limitations, this trend should persist in the coming years. Ideally, manufacturers increase their commitments to open-source hardware and software and create business models to support this change. Collaborations between industry and academia are luckily already prominent in the field and should continue to do so in the form of research projects, workshops, internships, field trials, and involvement in education [12].

APPENDIX A SUPPLEMENTARY MATERIALS

The supplementary materials can be found at: https://doi.org/ 10.5281/zenodo.7915562

- Supplement I. data extraction: UAVDataExtraction.xlsx
- Supplement II. variables explained: UAVDataExtractionSupportingDocument.pdf
- Supplement III. quality assessment: UAVQualityAssessmentExamples.xlsx
- Supplement IV. primary research: primary.xlsx
- Supplement V. secondary: secondary.xlsx

APPENDIX B GRAPHICAL OVERVIEW

See Figute 11.

APPENDIX C OVERVIEW OF SECONDARY STUDIES

See Table 12.

ACKNOWLEDGMENT

The authors acknowledge valuable help and contributions from all partners of the project.

REFERENCES

- C. Kislik, I. Dronova, and M. Kelly, "UAVs in support of algal Bloom research: A review of current applications and future opportunities," *Drones*, vol. 2, no. 4, p. 35, Oct. 2018.
- [2] R. Kellermann, T. Biehle, and L. Fischer, "Drones for parcel and passenger transportation: A literature review," *Transp. Res. Interdiscipl. Perspect.*, vol. 4, Mar. 2020, Art. no. 100088.
- [3] I. L. H. Alsammak, M. A. Mahmoud, H. Aris, M. AlKilabi, and M. N. Mahdi, "The use of swarms of unmanned aerial vehicles in mitigating area coverage challenges of forest-fire-extinguishing activities: A systematic literature review," *Forests*, vol. 13, no. 5, p. 811, May 2022.
- [4] F. Nex, C. Armenakis, M. Cramer, D. A. Cucci, M. Gerke, E. Honkavaara, A. Kukko, C. Persello, and J. Skaloud, "UAV in the advent of the twenties: Where we stand and what is next," *ISPRS J. Photogramm. Remote Sens.*, vol. 184, pp. 215–242, Feb. 2022.
- [5] C.-U. Hyun, M. Park, and W. Y. Lee, "Remotely piloted aircraft system (RPAS)-based wildlife detection: A review and case studies in maritime Antarctica," *Animals*, vol. 10, no. 12, p. 2387, Dec. 2020.
- [6] D. C. Tsouros, S. Bibi, and P. G. Sarigiannidis, "A review on UAVbased applications for precision agriculture," *Information*, vol. 10, no. 11, p. 349, Nov. 2019.
- [7] R. Dainelli, P. Toscano, S. F. Di Gennaro, and A. Matese, "Recent advances in unmanned aerial vehicles forest remote sensing—A systematic review—Part II: Research applications," *Forests*, vol. 12, no. 4, p. 397, Mar. 2021.
- [8] M. A. H. Zailani, R. Z. A. R. Sabudin, R. A. Rahman, I. M. Saiboon, A. Ismail, and Z. A. Mahdy, "Drone for medical products transportation in maternal healthcare: A systematic review and framework for future research," *Medicine*, vol. 99, no. 36, Sep. 2020, Art. no. e21967.

- [9] B. Kitchenham, R. Pretorius, D. Budgen, O. P. Brereton, M. Turner, M. Niazi, and S. Linkman, "Systematic literature reviews in software engineering—A tertiary study," *Inf. Softw. Technol.*, vol. 52, no. 8, pp. 792–805, Aug. 2010.
- [10] B. A. Kitchenham and S. Charters, "Guidelines for performing systematic literature reviews in software engineering," Version 2.3., Software Eng. Group, School Comput. Sci. Math., Dept. Comput. Sci., Keele Univ., Univ. Durham, Keele, U.K., Tech. Rep. EBSE-2007-01, 2007.
- [11] NIHR Centre for Reviews and Dissemination. (2015). About DARE. [Online]. Available: https://www.crd.york.ac.uk/CRDWeb/ AboutPage.asp
- [12] Z. Kotti, R. Galanopoulou, and D. Spinellis, "Machine learning for software engineering: A tertiary study," ACM Comput. Surv., vol. 55, no. 12, pp. 1–39, Dec. 2023.
- [13] S. Kim and K. Kim, "Systematic tertiary study for consolidating further implications of unmanned aircraft system applications," *J. Manage. Eng.*, vol. 37, no. 2, Mar. 2021, Art. no. 03120001.
- [14] K. Anderson and K. J. Gaston, "Lightweight unmanned aerial vehicles will revolutionize spatial ecology," *Frontiers Ecol. Environ.*, vol. 11, no. 3, pp. 138–146, Apr. 2013.
- [15] I. Colomina and P. Molina, "Unmanned aerial systems for photogrammetry and remote sensing: A review," *ISPRS J. Photogramm. Remote Sens.*, vol. 92, pp. 79–97, Jun. 2014.
- [16] M. J. Page et al., "The PRISMA 2020 statement: An updated guideline for reporting systematic reviews," *Syst. Rev.*, vol. 10, no. 1, pp. 1–11, Dec. 2021.
- [17] D. Budgen, P. Brereton, S. Drummond, and N. Williams, "Reporting systematic reviews: Some lessons from a tertiary study," *Inf. Softw. Technol.*, vol. 95, pp. 62–74, Mar. 2018.
- [18] T. Villa, F. Gonzalez, B. Miljievic, Z. Ristovski, and L. Morawska, "An overview of small unmanned aerial vehicles for air quality measurements: Present applications and future prospectives," *Sensors*, vol. 16, no. 7, p. 1072, Jul. 2016.
- [19] A. Ampatzoglou, S. Bibi, P. Avgeriou, M. Verbeek, and A. Chatzigeorgiou, "Identifying, categorizing and mitigating threats to validity in software engineering secondary studies," *Inf. Softw. Technol.*, vol. 106, pp. 201–230, Feb. 2019.
- [20] R. Luppicini and A. So, "A technoethical review of commercial drone use in the context of governance, ethics, and privacy," *Technol. Soc.*, vol. 46, pp. 109–119, Aug. 2016.
- [21] R. Merkert and J. Bushell, "Managing the drone revolution: A systematic literature review into the current use of airborne drones and future strategic directions for their effective control," J. Air Transp. Manage., vol. 89, Oct. 2020, Art. no. 101929.
- [22] F. C. Eugenio, C. T. Schons, C. L. Mallmann, M. S. Schuh, P. Fernandes, and T. L. Badin, "Remotely piloted aircraft systems and forests: A global state of the art and future challenges," *Can. J. Forest Res.*, vol. 50, no. 8, pp. 705–716, Aug. 2020. [Online]. Available: https://www.scopus.com/inward/record.uri?eid=2-s2.0-85088893036
- [23] B. Arshad, R. Ogie, J. Barthelemy, B. Pradhan, N. Verstaevel, and P. Perez, "Computer vision and IoT-based sensors in flood monitoring and mapping: A systematic review," *Sensors*, vol. 19, no. 22, p. 5012, Nov. 2019.
- [24] Y. Mualla, A. Najjar, A. Daoud, S. Galland, C. Nicolle, A.-U.-H. Yasar, and E. Shakshuki, "Agent-based simulation of unmanned aerial vehicles in civilian applications: A systematic literature review and research directions," *Future Gener. Comput. Syst.*, vol. 100, pp. 344–364, Nov. 2019.
- [25] M. R. B. Júnior, B. R. D. A. Moreira, A. L. D. B. Filho, D. Tedesco, L. S. Shiratsuchi, and R. P. D. Silva, "UAVs to monitor and manage sugarcane: Integrative review," *Agronomy*, vol. 12, no. 3, p. 661, Mar. 2022.
- [26] Ö. Dündar, M. Bilici, and T. Ünler, "Design and performance analyses of a fixed wing battery VTOL UAV," *Eng. Sci. Technol., Int. J.*, vol. 23, no. 5, pp. 1182–1193, Oct. 2020.
- [27] J. Huijbregts. (Oct. 2016). Intel Brengt Octacopter Voor Industriële Toepassingen Onder Eigen Naam Uit. [Online]. Available: https://tweakers.net/nieuws/116667/intel-brengt-octacopter-voorindustriele-toepassingen-onder-eigen-naam-uit.html
- [28] S. Park and Y. Choi, "Applications of unmanned aerial vehicles in mining from exploration to reclamation: A review," *Minerals*, vol. 10, no. 8, p. 663, Jul. 2020.

- [29] N. Sliusar, T. Filkin, M. Huber-Humer, and M. Ritzkowski, "Drone technology in municipal solid waste management and landfilling: A comprehensive review," *Waste Manage.*, vol. 139, pp. 1–16, Feb. 2022.
- [30] E. V. Butilă and R. G. Boboc, "Urban traffic monitoring and analysis using unmanned aerial vehicles (UAVs): A systematic literature review," *Remote Sens.*, vol. 14, no. 3, p. 620, Jan. 2022.
- [31] Y. Diez, S. Kentsch, M. Fukuda, M. L. L. Caceres, K. Moritake, and M. Cabezas, "Deep learning in forestry using UAV-acquired RGB data: A practical review," *Remote Sens.*, vol. 13, no. 14, p. 2837, Jul. 2021.
- [32] F. Librán-Embid, F. Klaus, T. Tscharntke, and I. Grass, "Unmanned aerial vehicles for biodiversity-friendly agricultural landscapes—A systematic review," *Sci. Total Environ.*, vol. 732, Aug. 2020, Art. no. 139204.
- [33] S. Macenski, T. Foote, B. Gerkey, C. Lalancette, and W. Woodall, "Robot operating system 2: Design, architecture, and uses in the wild," *Sci. Robot.*, vol. 7, no. 66, May 2022, Art. no. eabm6074.
- [34] G. Yang, J. Liu, C. Zhao, Z. Li, Y. Huang, H. Yu, B. Xu, X. Yang, D. Zhu, X. Zhang, R. Zhang, H. Feng, X. Zhao, Z. Li, H. Li, and H. Yang, "Unmanned aerial vehicle remote sensing for field-based crop phenotyping: Current status and perspectives," *Frontiers Plant Sci.*, vol. 8, pp. 1–28, Jun. 2017.
- [35] L. G. Poley and G. J. McDermid, "A systematic review of the factors influencing the estimation of vegetation aboveground biomass using unmanned aerial systems," *Remote Sens.*, vol. 12, no. 7, p. 1052, Mar. 2020.
- [36] J. Lin, M. Wang, M. Ma, and Y. Lin, "Aboveground tree biomass estimation of sparse subalpine coniferous forest with UAV oblique photography," *Remote Sens.*, vol. 10, no. 11, p. 1849, Nov. 2018.
- [37] C. Stöcker, R. Bennett, F. Nex, M. Gerke, and J. Zevenbergen, "Review of the current state of UAV regulations," *Remote Sens.*, vol. 9, no. 5, p. 459, May 2017.
- [38] B. Schäffer, R. Pieren, K. Heutschi, J. M. Wunderli, and S. Becker, "Drone noise emission characteristics and noise effects on humans— A systematic review," *Int. J. Environ. Res. Public Health*, vol. 18, no. 11, p. 5940, Jun. 2021.
- [39] A. Gohari, A. B. Ahmad, R. B. A. Rahim, A. Supa'at, S. A. Razak, and M. S. M. Gismalla, "Involvement of surveillance drones in smart cities: A systematic review," *IEEE Access*, vol. 10, pp. 56611–56628, 2022.
- [40] M. H. M. Ghazali, K. Teoh, and W. Rahiman, "A systematic review of real-time deployments of UAV-based LoRa communication network," *IEEE Access*, vol. 9, pp. 124817–124830, 2021.
- [41] S. M. Daud, M. M. Yusof, C. Heo, L. Khoo, M. C. Singh, M. Mahmood, and H. Nawawi, "Applications of drone in disaster management: A scoping review," *Sci. Justice*, vol. 62, no. 1, pp. 30–42, 2022.
- [42] R. M. Carrillo-Larco, M. Moscoso-Porras, A. Taype-Rondan, A. Ruiz-Alejos, and A. Bernabe-Ortiz, "The use of unmanned aerial vehicles for health purposes: A systematic review of experimental studies," *Global Health, Epidemiol. Genomics*, vol. 3, p. e13, Jan. 2018.
- [43] P. C. P. Júnior, A. Monteiro, R. D. L. Ribeiro, A. C. Sobieranski, and A. von Wangenheim, "Comparison of supervised classifiers and image features for crop rows segmentation on aerial images," *Appl. Artif. Intell.*, vol. 34, no. 4, pp. 271–291, Mar. 2020.
- [44] M. Coccia, "Sources of technological innovation: Radical and incremental innovation problem-driven to support competitive advantage of firms," *Technol. Anal. Strategic Manage.*, vol. 29, no. 9, pp. 1048–1061, Oct. 2017.
- [45] (2022). Innovation Highlight: Toroidal Propeller. Lincol Lab. MIT, Cambridge, MA, USA. [Online]. Available: https://www.ll.mit. edu/sites/default/iles/other/doc/2023-02/TVO_Technology_Highlight_41 _Toroidal_Propeller.pdf
- [46] E. Small, E. Fresk, and G. Nikolakopoulos, "Uav_Comm: A generalized communication between the robot operating system and low-level flight controllers," in *Proc. 28th Medit. Conf. Control Autom. (MED)*, Sep. 2020, pp. 1027–1032.
- [47] J. L. Tokar, "A comparison of avionics open system architectures," ACM SIGAda Ada Lett., vol. 36, no. 2, pp. 22–26, May 2017.
- [48] D. B. M. Yousefi, A. S. M. Rafie, S. A. R. Al-Haddad, and S. Azrad, "A systematic literature review on the use of deep learning in precision livestock detection and localization using unmanned aerial vehicles," *IEEE Access*, vol. 10, pp. 80071–80091, 2022.
- [49] M. H. Widianto, M. I. Ardimansyah, H. I. Pohan, and D. R. Hermanus, "A systematic review of current trends in artificial intelligence for smart farming to enhance crop yield," *J. Robot. Control (JRC)*, vol. 3, no. 3, pp. 269–278, May 2022.

- [50] A. Abdollahi, B. Pradhan, N. Shukla, S. Chakraborty, and A. Alamri, "Deep learning approaches applied to remote sensing datasets for road extraction: A state-of-the-art review," *Remote Sens.*, vol. 12, no. 9, p. 1444, May 2020.
- [51] P. Muruganantham, S. Wibowo, S. Grandhi, N. H. Samrat, and N. Islam, "A systematic literature review on crop yield prediction with deep learning and remote sensing," *Remote Sens.*, vol. 14, no. 9, p. 1990, Apr. 2022.
- [52] D. Tuia, B. Kellenberger, S. Beery, B. R. Costelloe, S. Zuffi, B. Risse, A. Mathis, M. W. Mathis, F. van Langevelde, T. Burghardt, R. Kays, H. Klinck, M. Wikelski, I. D. Couzin, G. van Horn, M. C. Crofoot, C. V. Stewart, and T. Berger-Wolf, "Perspectives in machine learning for wildlife conservation," *Nature Commun.*, vol. 13, no. 1, p. 792, Feb. 2022.
- [53] T. Urakubo and A. Budiyono, "Advancements in tilted-rotor unmanned aerial vehicles: A comprehensive review," J. Instrum., Autom. Syst., vol. 10, no. 3, pp. 92–104, 2023. [Online]. Available: http://ojs. unsysdigital.com/index.php/jias/article/view/1150
- [54] G. von Krogh and S. Spaeth, "The open source software phenomenon: Characteristics that promote research," J. Strategic Inf. Syst., vol. 16, no. 3, pp. 236–253, Sep. 2007.
- [55] L. Meier, P. Tanskanen, F. Fraundorfer, and M. Pollefeys, "PIXHAWK: A system for autonomous flight using onboard computer vision," in *Proc. IEEE Int. Conf. Robot. Autom.*, May 2011, pp. 2992–2997.
- [56] ArduPilot Development Team. (2023). Mission Planner. [Online]. Available: https://ardupilot.org/planner/
- [57] C. Griwodz, S. Gasparini, L. Calvet, P. Gurdjos, F. Castan, B. Maujean, G. De Lillo, and Y. Lanthony, "AliceVision meshroom: An open-source 3D reconstruction pipeline," in *Proc. 12th ACM Multimedia Syst. Conf.* New York, NY, USA: ACM Press, Jun. 2021, pp. 241–247.
- [58] OpenDroneMap Authors. (2020). ODM-A Command Line Toolkit To Generate Maps, Point Clouds, 3D Models and DEMs From Drone, Ballon or Kite Images. [Online]. Available: https://github.com/OpenDroneMap/ODM
- [59] QGIS Development Team. (2023). QGIS Geographic Information System. [Online]. Available: https://www.qgis.org
- [60] GDAL/OGR Contributors. (2022). GDAL/OGR Geospatial Data Abstraction Software Library. [Online]. Available: https://gdal.org
- [61] A. Paszke et al., "PyTorch: An imperative style, high-performance deep learning library," in *Proc. Adv. Neural Inf. Process. Syst.* Red Hook, NY, USA: Curran Associates, Dec. 2019, pp. 8024–8035. [Online]. Available: http://papers.neurips.cc/paper/9015-pytorch-an-imperativestyle-high-performance-deep-learning-library.pdf
- [62] F. Chollet. (2015). Keras. [Online]. Available: https://keras.io
- [63] M. Abadi et al. (2015). TensorFlow: Large-Scale Machine Learning on Heterogeneous Systems. [Online]. Available: https://www.tensorflow.org/
- [64] S. M. Jameii, R. S. Zamirnaddafi, and R. Rezabakhsh, "Internet of flying things security: A systematic review," *Concurrency Comput., Pract. Exper.*, vol. 34, no. 24, pp. 1–15, Nov. 2022.
- [65] N. Fatima, P. Saxena, and M. Gupta, "Integration of multi access edge computing with unmanned aerial vehicles: Current techniques, open issues and research directions," *Phys. Commun.*, vol. 52, Jun. 2022, Art. no. 101641.
- [66] L. Hernandez-Santin, M. L. Rudge, R. E. Bartolo, and P. D. Erskine, "Identifying species and monitoring understorey from UAS-derived data: A literature review and future directions," *Drones*, vol. 3, no. 1, p. 9, Jan. 2019.
- [67] K. Dushime, L. Nkenyereye, S. K. Yoo, and J. Song, "A review on collision avoidance systems for unmanned aerial vehicles," in *Proc. Int. Conf. Inf. Commun. Technol. Converg. (ICTC)*, Oct. 2021, pp. 1150–1155.
- [68] Z. Qadir, F. Ullah, H. S. Munawar, and F. Al-Turjman, "Addressing disasters in smart cities through UAVs path planning and 5G communications: A systematic review," *Comput. Commun.*, vol. 168, pp. 114–135, Feb. 2021.
- [69] M. Mohan, G. Richardson, G. Gopan, M. M. Aghai, S. Bajaj, G. A. P. Galgamuwa, M. Vastaranta, P. S. P. Arachchige, L. Amorós, A. P. D. Corte, S. De-Miguel, R. V. Leite, M. Kganyago, E. N. Broadbent, W. Doaemo, M. A. B. Shorab, and A. Cardil, "UAV-supported forest regeneration: Current trends, challenges and implications," *Remote Sens.*, vol. 13, no. 13, p. 2596, Jul. 2021.
- [70] M. Kucharczyk and C. H. Hugenholtz, "Remote sensing of natural hazard-related disasters with small drones: Global trends, biases, and research opportunities," *Remote Sens. Environ.*, vol. 264, Oct. 2021, Art. no. 112577.

- [71] M. Sibanda, O. Mutanga, V. G. P. Chimonyo, A. D. Clulow, C. Shoko, D. Mazvimavi, T. Dube, and T. Mabhaudhi, "Application of drone technologies in surface water resources monitoring and assessment: A systematic review of progress, challenges, and opportunities in the global south," *Drones*, vol. 5, no. 3, p. 84, Aug. 2021.
- [72] M. Mohamed and D. Q. Tran, "Content analysis of e-inspection implementation for highway infrastructure construction projects," *Eng.*, *Construct. Architectural Manage.*, vol. 30, no. 7, pp. 2621–2644, Aug. 2023.
- [73] H. Pathak, C. Igathinathane, Z. Zhang, D. Archer, and J. Hendrickson, "A review of unmanned aerial vehicle-based methods for plant stand count evaluation in row crops," *Comput. Electron. Agricult.*, vol. 198, Jul. 2022, Art. no. 107064.
- [74] T. S. N. Rachmawati and S. Kim, "Unmanned aerial vehicles (UAV) integration with digital technologies toward Construction 4.0: A systematic literature review," *Sustainability*, vol. 14, no. 9, p. 5708, May 2022.
- [75] F.-L. Chiper, A. Martian, C. Vladeanu, I. Marghescu, R. Craciunescu, and O. Fratu, "Drone detection and defense systems: Survey and a softwaredefined radio-based solution," *Sensors*, vol. 22, no. 4, p. 1453, Feb. 2022.
- [76] R. Shadiev and S. Yi, "A systematic review of UAV applications to education," *Interact. Learn. Environments*, vol. 31, no. 10, pp. 6165–6194, Dec. 2023.
- [77] S. Zhou and M. Gheisari, "Unmanned aerial system applications in construction: A systematic review," *Construct. Innov.*, vol. 18, no. 4, pp. 453–468, Nov. 2018.
- [78] A. I. Novikov and B. T. Ersson, "Aerial seeding of forests in russia: A selected literature analysis," *IOP Conf. Ser., Earth Environ. Sci.*, vol. 226, Feb. 2019, Art. no. 012051.
- [79] H. Vanderhorst, S. Suresh, and R. Suresh, "Systematic literature research of the current implementation of unmanned aerial system (UAS) in the construction industry," *Int. J. Innov. Technol. Exploring*, vol. 8, no. 11, pp. 416–428, 2019.
- [80] K. Haula and E. Agbozo, "A systematic review on unmanned aerial vehicles in sub-Saharan Africa: A socio-technical perspective," *Technol. Soc.*, vol. 63, Nov. 2020, Art. no. 101357.
- [81] C. Zhang, J. Valente, L. Kooistra, L. Guo, and W. Wang, "Orchard management with small unmanned aerial vehicles: A survey of sensing and analysis approaches," *Precis. Agricult.*, vol. 22, no. 6, pp. 2007–2052, Dec. 2021.
- [82] N. Amarasingam, A. S. A. Salgadoe, K. Powell, L. F. Gonzalez, and S. Natarajan, "A review of UAV platforms, sensors, and applications for monitoring of sugarcane crops," *Remote Sens. Appl., Soc. Environ.*, vol. 26, Apr. 2022, Art. no. 100712.
- [83] A. Duarte, N. Borralho, P. Cabral, and M. Caetano, "Recent advances in forest insect pests and diseases monitoring using UAV-based data: A systematic review," *Forests*, vol. 13, no. 6, p. 911, Jun. 2022.
- [84] U. Iqbal, P. Perez, W. Li, and J. Barthelemy, "How computer vision can facilitate flood management: A systematic review," *Int. J. Disaster Risk Reduction*, vol. 53, Feb. 2021, Art. no. 102030.
- [85] L. A. S. Cardoso, P. R. S. Farias, and J. A. C. Soares, "Use of unmanned aerial vehicle in sugarcane cultivation in Brazil: A review," *Sugar Tech*, vol. 24, no. 6, pp. 1636–1648, Dec. 2022.
- [86] H. T. Berie and I. Burud, "Application of unmanned aerial vehicles in Earth resources monitoring: Focus on evaluating potentials for forest monitoring in Ethiopia," *Eur. J. Remote Sens.*, vol. 51, no. 1, pp. 326–335, Jan. 2018.
- [87] J. S. Álvares and D. B. Costa, "Literature review on visual construction progress monitoring using unmanned aerial vehicles," in *Proc. 26th Annu. Conf. Int. Group Lean Construct.* Chennai, India: The International Group for Lean Construction, Jul. 2018, pp. 669–680.
- [88] G. Albeaino, M. Gheisari, and B. W. Franz, "A systematic review of unmanned aerial vehicle application areas and technologies in the AEC domain," *J. Inf. Technol. Construct.*, vol. 24, pp. 381–405, Jul. 2019.
- [89] N. Rebolo-Ifrán, M. Graña Grilli, and S. A. Lambertucci, "Drones as a threat to wildlife: Youtube complements science in providing evidence about their effect," *Environ. Conservation*, vol. 46, no. 3, pp. 205–210, Sep. 2019.
- [90] L. Luo, X. Wang, H. Guo, R. Lasaponara, X. Zong, N. Masini, G. Wang, P. Shi, H. Khatteli, F. Chen, S. Tariq, J. Shao, N. Bachagha, R. Yang, and Y. Yao, "Airborne and spaceborne remote sensing for archaeological and cultural heritage applications: A review of the century (1907–2017)," *Remote Sens. Environ.*, vol. 232, Oct. 2019, Art. no. 111280.

- [91] C. Dupuis, P. Lejeune, A. Michez, and A. Fayolle, "How can remote sensing help monitor tropical moist forest degradation? A systematic review," *Remote Sens.*, vol. 12, no. 7, p. 1087, Mar. 2020.
- [92] A. Thibbotuwawa, G. Bocewicz, P. Nielsen, and Z. Banaszak, "Unmanned aerial vehicle routing problems: A literature review," *Appl. Sci.*, vol. 10, no. 13, p. 4504, Jun. 2020.
- [93] F. Elghaish, S. Matarneh, S. Talebi, M. Kagioglou, M. R. Hosseini, and S. Abrishami, "Toward digitalization in the construction industry with immersive and drones technologies: A critical literature review," *Smart Sustain. Built Environ.*, vol. 10, no. 3, pp. 345–363, Nov. 2021.
- [94] R. Dainelli, P. Toscano, S. F. Di Gennaro, and A. Matese, "Recent advances in unmanned aerial vehicle forest remote sensing—A systematic review—Part I: A general framework," *Forests*, vol. 12, no. 3, p. 327, Mar. 2021.
- [95] G.-G. Hognogi, A.-M. Pop, A.-C. Marian-Potra, and T. Somesfălean, "The role of UAS-GIS in digital era governance. A systematic literature review," *Sustainability*, vol. 13, no. 19, p. 11097, Oct. 2021.
- [96] J. Sinnemann, M. Boshoff, R. Dyrska, S. Leonow, M. Mönnigmann, and B. Kuhlenkötter, "Systematic literature review of applications and usage potentials for the combination of unmanned aerial vehicles and mobile robot manipulators in production systems," *Prod. Eng.*, vol. 16, no. 5, pp. 579–596, Oct. 2022.
- [97] S. Ecke, J. Dempewolf, J. Frey, A. Schwaller, E. Endres, H.-J. Klemmt, D. Tiede, and T. Seifert, "UAV-based forest health monitoring: A systematic review," *Remote Sens.*, vol. 14, no. 13, p. 3205, Jul. 2022.
- [98] D. R. Viloria, E. L. Solano-Charris, A. Muñoz-Villamizar, and J. R. Montoya-Torres, "Unmanned aerial vehicles/drones in vehicle routing problems: A literature review," *Int. Trans. Oper. Res.*, vol. 28, no. 4, pp. 1626–1657, Jul. 2021.
- [99] M. Stampa, A. Sutorma, U. Jahn, J. Thiem, C. Wolff, and C. Röhrig, "Maturity levels of public safety applications using unmanned aerial systems: A review," *J. Intell. Robotic Syst.*, vol. 103, no. 1, pp. 1–15, Sep. 2021.
- [100] S. Mirri, C. Prandi, and P. Salomoni, "Human-drone interaction: State of the art, open issues and challenges," in *Proc. ACM SIGCOMM Workshop Mobile AirGround Edge Comput., Syst., Netw., Appl.* New York, NY, USA: Association for Computing Machinery, Aug. 2019, pp. 43–48.
- [101] S. O. Johnsen and T. E. Evjemo, "State of the art of unmanned aircraft transport systems in industry related to risks, vulnerabilities and improvement of safety," in *Proc. 29th Eur. Saf. Rel. Conf. (ESREL)*, Z. E. Beer, Ed. Hannover, Germany: Research Publishing Services, 2019, pp. 4154–4161.
- [102] S. S. Moumgiakmas, G. G. Samatas, and G. A. Papakostas, "Computer vision for fire detection on UAVs—From software to hardware," *Future Internet*, vol. 13, no. 8, p. 200, Jul. 2021.
- [103] J. Kaivosoja, J. Hautsalo, J. Heikkinen, L. Hiltunen, P. Ruuttunen, R. Näsi, O. Niemeläinen, M. Lemsalu, E. Honkavaara, and J. Salonen, "Reference measurements in developing UAV systems for detecting pests, weeds, and diseases," *Remote Sens.*, vol. 13, no. 7, p. 1238, Mar. 2021.
- [104] X. Lyu, X. Li, D. Dang, H. Dou, K. Wang, and A. Lou, "Unmanned aerial vehicle (UAV) remote sensing in grassland ecosystem monitoring: A systematic review," *Remote Sens.*, vol. 14, no. 5, p. 1096, Feb. 2022.
- [105] F. Ahmed, J. Mohanta, A. Keshari, and P. S. Yadav, "Recent advances in unmanned aerial vehicles: A review," *Arabian J. Sci. Eng.*, vol. 47, no. 7, pp. 7963–7984, 2022.



KWABENA E. BENNIN received the B.A. degree (Hons.) in computer science and statistics from the University of Ghana and the Ph.D. degree in computer science from the City University of Hong Kong. He was a Software Applications Developer with Vodafone Ghana for two years. He is currently an Assistant Professor with the Information Technology Group. Prior to working with WUR, he was a Postdoctoral Researcher with the Department of Software Engineering, Blekinge

Institute of Technology, and a Data Science Consultant with Ericsson Karlskrona, Sweden. He has vast experience as a Software Engineer. He has more than 40 international publications in the areas of empirical software engineering, software testing, and data analytics. His research focuses on empirical software engineering, software quality, and software analytics. His current research interest includes the application of AI for software engineering tasks.



ÖNDER BABUR received the B.Sc. degree from METU, Turkey, the M.Sc. degree from RWTH Aachen University, Germany, and the Ph.D. degree from the Eindhoven University of Technology. He was employed as a Postdoctoral Researcher with the Software Engineering & Technology (SET) Group, Eindhoven University of Technology, from 2019 to 2021. He has further experience as a Software Engineer in Germany and a Researcher in Spain. Over the years, he has

participated in a number of research projects on automotive software engineering, digitalization and industrial automation, and multiscale modeling. He has initiated and co-chaired the International Workshop on Analytics and Mining of Model Repositories and the International Conference on Systems Modelling and Management. Around the main research lines, he has been collaborating with many international research groups and hightech companies. His main research interests include the fields of modeldriven engineering, systems modeling, software analytics, AI for software engineering, and empirical software engineering.



JURRIAN DOORNBOS received the M.Sc. degree in geo-information science with a specialization in low-cost hardware and sustainability from Wageningen University & Research, where he is currently pursuing the Ph.D. degree with the Information Technology Group. His current research interest includes open-source hardware and software for drone applications with a focus on the generalization of AI using a generative approach.



JOÃO VALENTE (Member, IEEE) received the Ph.D. degree (cum laude) in robotics and automation from the Polytechnic University of Madrid (UPM), in 2014. He is currently an Assistant Professor with the Information Technology Group (INF), Wageningen University & Research (WUR). He has been with unmanned aerial vehicles (UAVs) and field robotics for more than 15 years. He has dedicated his career to combining these technologies to develop robust solutions for

Precision Agriculture and Ecology Management. He is the author of more than 50 publications on this topic. His research interests include UAVs, robotics, artificial intelligence (AI), and remote sensing. He was awarded three times for the best Ph.D. thesis in the academic and industrial sectors.